Fostering Non-Cognitive Skills in Active Labor Market Programs: Evidence from a large-scale RCT *

Analia Schlosser, Tel Aviv University

Yannay Shanan, Bar Ilan University

April 2021

Abstract

The long-term unemployed sometimes lack basic non-cognitive skills needed to enter and succeed in the labor market. We examine whether it is possible to develop or enhance these skills among adults by using a large-scale randomized control trial (RCT) to evaluate the effectiveness of an Active Labor Market Program (ALMP) that targets income-support claimants in Israel. In this program, participants receive personalized treatment composed of weekly sessions with occupational trainers and motivational group workshops. We find that the program increased the participants' employment rate by 7.9 percentage points and decreased income support recipiency by 10.5 percentage points relative to the control group. The effects are larger among high-school dropouts and those with a longer history of welfare dependence. The program also boosted the employment of participants' non-treated husbands but had no effect on participants' non-treated wives. There is no evidence of displacement effects on the control group. Analysis of the mechanisms at work shows that the program has positive and significant effects on its participants' work self-efficacy and job-search self-efficacy. We conclude that unemployed income-support claimants with lower labor-force attachment can benefit considerably from interventions that aim to improve their non-cognitive work-related skills.

^{*} We thank seminar participants at Freie Universität Berlin, Aalto University, IDC, Bar-Ilan University, Haifa University, IZA European Summer School in Labor Economics, Maastricht University, Inequality and Social Mobility Conference, and Advances in Field Experiments Conference. We thank the staff from the research department at the Israeli Employment Service for providing the administrative data on experminet's participants. This research was conducted at the research room of the National Insurance Institute of Israel (NII). We thank the NII research unit for providing the administrative data. Schlosser acknowledges financial support from the Pinhas Sapir Center for Development, the Foerder Institute for Economic Research, the National Insurance Institute of Israel, and the Pinhas Sapir Economic Policy Forum. AEA Trial Registry RCT ID: AEARCTR-0007548.

I. Introduction

Active Labor Market Programs (ALMPs) include a set of policies that aim to enhance the employability and earning capacity of individuals who are unemployed or on welfare. One of the most prevalent types of ALMPs are training programs (in traditional classrooms or on the job) that provide unemployed individuals with general skills or specific occupational skills in order to enhance their productivity and employability. Many such individuals, however, lack basic non-cognitive skills such as motivation, career aspirations, and interpersonal skills that are needed to transition from welfare to work and persevere in employment—skills that strongly predict labor-market success (see e.g. Heckman et al., 2006). Scientific evidence of the possibility of improving these skills, especially among the adult population, is limited, and little is known about the impact of such an improvement on labor-market outcomes and welfare dependence.

In this paper, we examine whether fostering welfare recipients' non-cognitive skills can enhance their likelihood of employment and subsequent earnings. To do this, we use a large-scale randomized control trial (RCT) to evaluate the effectiveness of an ALMP implemented in Israel. The program is designed to integrate chronically unemployed income-support claimants aged 20–50 into the labor force, preventing welfare dependency and long-term chronic unemployment. Its main goal is to foster participants' work-related non-cognitive skills such as motivation, work self-efficacy, self-esteem, and interpersonal skills. Individuals randomly assigned to the program receive individual coaching and participate in therapeutic groups for three to seven months, receiving also job search assistance. Overall, 48,000 individuals were allocated to the program from its inception in March 2014 to December 2018. Our paper focuses on the population allocated into the treated and control groups during the first year of the program implementation as a RCT: 6,151 individuals.

We combine administrative datasets from the Israeli Employment Service and Social Security records on employment, earnings, welfare, and disability benefits together with survey data to build a comprehensive picture of the individuals before, during, and after their allocation into treatment and control groups. Our main results show that twelve months after randomization, the program raised the participants' employment rates by 8 percentage points relative to the control group (a 24% increase), lessened their welfare dependency by 11 percentage points (a 26% decline), and lowered the share of treated participants reporting to the employment office by 15 percentage points (a 38% reduction). These effects persisted even eighteen months after allocation to the program. The impact of the program was greater among high-school dropouts and those with lower labor-force attachment, a longer history of

2

income-support recipiency, or self-reported health limitations. The program had spillover effects on treated women's husbands, increasing their employment rates by 6 percentage points (an increase of 12 percent), but there is no equivalent effect on the wives of treated men. There is also no evidence on externalities among the control group.

We find that the program worked through two different channels among different individuals: it generated a threat effect for some participants, inducing them to stop reporting to the employment office soon after their allocation to the treated group due to the additional burden of the program's requirements. Other participants benefited from the tools imparted by program, experiencing a significant increase in employment. The treatment effect on participants' non-cognitive skills is estimated as significant and positive with respect to self-reported work self-efficacy and job-search self-efficacy and marginally, on self-esteem, all of which are highly associated with labor-market success. The savings on welfare transfers offset the per-participant costs within twelve months.

Our study is related to a large literature that evaluates the effects of ALMPs. While most of the earlier studies were based on non-experimental data, the share of studies based on RCTs is increasing over time (see recent reviews by Kluve, 2010, and Card et al., 2018; and earlier work by Greenberg, 2003; and Greenberg et al., 2005).¹ ALMPs vary not only in their target populations and the local socioeconomic conditions they face but also in their approach toward the best way to tackle unemployment. The evidence suggests substantial heterogeneity among the effects of different programs. Kluve (2010), for example, finds that programs that focus on counseling and monitoring, job-search assistance (JSA), and corresponding sanctions in case of noncompliance outperform programs that focus on human capital enhancing measures, private-sector-incentive schemes, and direct employment. Card et al. (2018), find that job-search assistance and sanction programs have relatively large short-term impacts whereas training and private-sector employment programs have smaller short-term impacts but larger effects in the medium and longer run. In general, there is a wide consensus that public sector employment subsidies have negligible or negative impacts. 40% of the studies reviewed by Card et al. (2018) reported positive significant effects in the short-term and 61% did so for the longer term.

Overall, while some programs are found to be beneficial, less is known about why they work and under what circumstances. Several evaluations consider the possibility that participants in mandatory programs

¹ Kluve's (2010) meta-analysis, for example, includes only nine RCTs among 137 studies reviewed. Only one -fifth of the estimates reported by Card et al., (2018) are based on experimental studies.

may immediately forgo their claims and exit welfare or unemployment in order to avoid the additional "cost" associated with the program (Black et al., 2003, Dolton and O'Neill, 2002). This mechanism may explain the larger short-term effects of "work-first" programs. Other than this, the literature is rather silent about the underlying mechanisms of successful ALMPs. Remarkably, there is limited empirical evidence on programs that focus on enhancing non-cognitive skills among the unemployed. Recent developments in the literature that stress the importance of soft skills make research on these types of programs crucial. As Crépon and van den Berg (2016) point out, many unemployed individuals have been disconnected from the labor market for long periods and lack basic traits needed to reintegrate. Traditional ALMPs may be poorly designed for such reintegration. Instead, it might be important to focus on programs that boost participants' self-esteem and other personality traits through mentoring, therapy, and group treatments, in which similarly disadvantaged individuals stimulate each other.

Soft or non-cognitive skills, much like cognitive skills, can affect preferences, skill-formation technology, and productivity. Non-cognitive skills such as motivation, self-efficacy, and perseverance are found to be positively associated with test scores and labor-market outcomes (see Brunello and Schlotter, 2011, and Kautz et al., 2014, for a review of the literature). Moreover, several studies have found that the variance of many later-life outcomes explained by non-cognitive measures sometimes rivals that explained by measures of cognitive ability (see, e.g., Heckman et al., 2006, and Lindqvist and Vestman, 2011).

While personality traits and non-cognitive skills are relatively stable across situations, they are not necessarily permanent and some interventions can enhance them in lasting ways (Heckman and Kautz, 2012). Early-childhood programs such as Headstart and the Perry Preschool program were found to enhance non-cognitive skills and, consequently, promote higher social and economic success (Carneiro and Heckman, 2003; Kautz et al., 2014). There is scarce evidence, however, on returns to investments in non-cognitive skills later in life. A recent example of such evidence is given in a study on an intervention in Liberia among criminally engaged men (Blattman et al., 2017). The authors' findings imply that self-control and self-image are malleable in adults and that investments in enhancing these skills may mitigate crime and violence. Additional evidence is provided by Heller et al. (2017) who find a reduction in crime among disadvantaged youths in Chicago who participated in two different behavioral cognitive therapy programs. The authors however, did not find that the programs produced significant changes in participants' emotional intelligence or social skills, self-control or grit.

The contribution of the present study is threefold. First, it augments the literature that evaluates the effect of ALMPs on labor-market performance by examining its impacts using a clean experimental design, employing a large and heterogeneous sample, and not only analyzing standard labor-market outcomes but also examining the impact of the program on non-cognitive skills. It also provides a detailed evaluation of the dynamic effects of the intervention using administrative records on employment, earnings, and welfare recipiency before, during, and after allocation into treatment and control groups and provides a cost-benefit analysis of the program. Second, the study contributes to the growing literature that examines the development of non-cognitive skills and their importance for life outcomes by showing that some of these skills are malleable later in life and have an important role in enhancing employability of low-skilled individuals. Finally, we examine the impact of the intervention not only at the individual level but also at the household level including non-treated spouses, making this one of the first studies that illuminates spillover effects of ALMPs on other household members, demonstrating, importantly, that the benefits of these types of programs may be larger than previously thought.²

The remainder of the paper is organized as follows. Section II provides background on welfare support in Israel and describes the program and the experimental design. Section III presents the identification strategy. Section IV describes the data, defines the samples used throughout the study, and examines the effectiveness of the randomization. Section V reports the main estimates of program effect on a range of outcomes from administrative datasets, presents results on heterogeneous effects and dynamic treatment effects, and examines spillover effects on non-treated spouses and on the control group. Section VI explores the mechanisms that underlie the impact of the program, focusing on the mediating role of non-cognitive skills. Section VII concludes.

II. Background

Institutional Context and Description of the Program

The National Insurance Institute of Israel (NII) provides monthly income-support benefits to residents who cannot ensure themselves a basic minimum income for subsistence. In 2014, approximately 100,000 households, almost 5% of households countrywide, received such benefits. Eligibility for income support is based on age, income and, assets. Claimants who are considered capable of working (healthy, age below

² The only study we found that assessed spillovers effects of ALMPs on the household is Kugler et al. (forthcoming) who detected positive spillovers of a training program in Colombia on the likelihood of attaining tertiary education among participants' relatives.

sixty, and, among single parents, having children more than two years old) must report weekly (or monthly for those above age fifty) to one of seventy-five local employment offices run by the Israeli Employment Service (IES).³ Treatment at the employment office is minimal: individuals need to stamp their fingerprint in an automatic machine each week and meet once every three weeks (or when relevant) with a caseworker who gives them job referrals. Failure to report to the employment office or rejection of a relevant job offer results in denial of income-support payments. Working individuals who earn below a minimum amount set by law also receive income support; this is known as an *income supplement*. Incomesupplement recipients are not required to report to their local employment office every week. Instead, IES gives them time-limited exemptions, using discretion as to the duration of the exemptions and choosing whether to pursue a more demanding approach, for example, by requiring an increase in hours worked. Income-support recipients also receive reduced-cost services from other government ministries such as subsidized daycare, rent assistance, and a lower rate of property tax, in addition to the monthly income-support transfer. The maximal monthly transfer received by the head of household—a function of age, marital status, and number of dependent kin—ranged in 2014 between \$500 and \$1200 a month— 40% and 100% of the minimum wage, respectively.

In February 2014, IES launched an ALMP called "Employment Circles" in fourteen of its employment offices with the purpose of integrating unemployed income-support claimants into the labor force and preventing welfare dependency and long-term chronic unemployment. The target population were income-support claimants aged 20–50 who report to the employment office and are unemployed. The program focuses on enhancing participants' non-cognitive skills by providing personalized treatment composed of weekly sessions with occupational trainers, therapeutic group meetings with coaches, and job-search-assistance workshops. The program begins with two one-on-one meetings with an occupational trainer who diagnoses the participant in accordance with employability, motivational level, and barriers to employment, and recommends a specific track of group workshops and personal meetings on this basis. Together with the occupational trainer, participants define their career goals and build a program to attain them. A key component of the program is the group workshops, in which coaches focus on identifying participants' strengths; enhancing their motivation, job-search efficacy, work self-efficacy,

³ Exempt are prisoners currently performing community service or under house arrest, ex-prisoners during the first couple of months after their release, alcohol or drugs addicts, pregnant women, women in women's shelters, caregivers of a sick household member, and supervisors of a household member under house arrest.

and self-image; and developing a proactive work attitude.⁴ The workshops and the meetings with occupational trainers also focus on imparting skills conducive to secure stable employment, for example, by simulating workday situations and instilling basic concepts of work life along with training on job search skills. Appendix 1 elaborates on program content.

Unlike regular income-support claimants, who must report to the employment office once per week, program participants need to visit three times per week—twice for workshops and meetings with occupational trainers (3–5 hours) and once for a regular meeting with their caseworker. The program is mandatory, non-compliance leading to loss of income support. The program lasts two to seven months depending on the participant's specific needs. Participants can leave the program at any time if they find a job. In this case, they may continue to receive income-support benefits in the form of income supplement depending on the level of their labor income. After seven months, unemployed participants who still report to the employment office return to the regular track of weekly visits.

The program may increase its participants' employment and reduce their welfare dependence of its participants through different channels. First, the workshops and individual sessions may enhance their motivation, sense of job-search and work self-efficacy, and additional traits that may in turn affect job search, employment, and job persistency. A second and different channel is created by the additional requirement of the program to attend the employment office three times a week instead of once and the additional time that participants must spend there. These extra requirements raise the non-monetary costs of claiming welfare benefits. In addition, the extra attendance requirements at the employment office make it more difficult for one to work in the informal sector while declaring oneself unemployed and claiming benefits. While the program is not designed to test the contribution of each channel separately, we present below several bits of evidence that suggest that both channels are in place, affecting different groups of individuals.

Experimental Design

The program was implemented gradually using an experimental research design implemented in two waves. The first wave started on February 2014 in seven employment offices; a second wave including

⁴ The content of the training and the workshops is based on the STRIVE international model developed by Strive US (<u>https://strive.org/</u>), which emphasizes personal development and improvement of tools needed to integrate into and excel at a job. The model was adapted to and tested for the Israeli context by the Israeli employment incubator JDC-Tevet.

seven additional offices followed in August 2014.⁵ These fourteen offices constituted the experimental sample for the RCT. The program was then gradually expanded to include almost all employment offices countrywide and the age limit was raised to fifty-five. Table 1 reports some basic characteristics of the employment offices included in the RCT and all other employment offices. The experiment offices served roughly 45% of unemployed Israeli welfare claimants in 2014. The average jobseeker is thirty-eight years old, has no more than ten years of schooling, and is most likely a woman. Most claimants are also Arab, this population being substantially overrepresented in the Israeli welfare system. ⁶ Overall, the characteristics of the offices included in the experimental phase are highly similar to the remaining offices, both in terms of the population demographics and local labor-market conditions (summarized in this table by local unemployment rates and locality socioeconomic indices). This similarity supports the relevance of our findings for the program scale-up.

During the experimental phase of the program, individuals who submitted new income-support claims and a fraction of existing claimants in the welfare system were randomized into control and treatment groups. Randomization took place on a weekly basis separately for the incoming flow of jobseekers (i.e., new and returning claimants) and the stock of current jobseekers (the existing pool of claimants) at each employment office. The number of individuals assigned to treatment and control groups varied over time due to changes in the incoming flow of claimants and the capacity of the program at the office level. Randomization was achieved by a software protocol that was implemented on the premises of the IES research office to avoid manipulations. Treatment status was updated in the central IES operational database and the local employment offices received the list of individuals allocated to the treatment group. Treatment status was assigned at the household level. Namely, in cases where both partners attend the employment office, both were assigned to one group: treatment or control.⁷ In practice, as we will discuss later, in most cases only one household member was assigned to the program because the other partner was not registered with the employment office during the period we analyze. This allowed us to examine the effect of the program on non-treated spouses.

⁵ The employment offices chosen for the first waves of the program include the largest IES offices, collectively serving almost half of the Israeli population.

⁶ The Arab population accounts for about one-fifth of Israel's population.

⁷ This is also the case when only one partner is registered at the employment office on the allocation date but the other partner registers a few months later. If the jobseeker's partner was already assigned to treatment, his/her household treatment status is already registered and he/she is informed about his/her assignment to the program upon the next visit.

Upon their next visit to the employment office, treated individuals stamped their fingerprints at the automated machine and received a notification that required them to meet with a designated caseworker who informed them that they had been selected for the program. Individuals randomly assigned to the control group received no notification and continued to follow the usual protocol of a weekly visit to the employment office. An individual's treatment or control status was defined once and remained in effect even if he or she moved to another city, stopped reporting to the employment office, or re-registered with IES after a certain period.

III. Empirical Framework

Through the mechanism of randomization, we can infer the effect of the program by estimating the difference in post-program outcomes between the treatment and the control group after controlling for the randomization unit, thus averting the problem of selection bias.⁸ Accordingly, we estimate the average treatment effect of the program by regressing various outcomes on a treatment dummy while controlling for the randomization cell.⁹ A small fraction of the treatment group (around 1%) did not receive the services of the program for various reasons ranging from administrative errors to total exemption on grounds of serious physical- or mental-health issues.¹⁰ We include them in the treatment group to avoid selection. Therefore, we estimate the intention to treat effect. Given the negligible share of treatment-group members who were exempted from the program, we do not use an instrumental-variable strategy to estimate the treatment effect on the treated since we expect to obtain almost identical estimates in this case. To increase precision and to control for small differences between treated and control groups that derive from randomization in a finite sample, we augment the basic model with a vector of covariates that include individuals' demographic characteristics, employment, and welfare history measured before randomization. The estimating equation can be written as follows:

(1) $Outcome_{ijtp} = \beta Treatment_i + X_i' \varphi + \gamma_{itp} + \varepsilon_{ijtp}$

⁸ This is under the assumption that the program has no externalities to the control group. We assess this assumption in Section V and Appendix 2 below.

⁹ We aggregate the randomization cell at the month level instead of the week to avoid cases of singletons and enhance precision. In practice, the estimates are virtually identical in both cases.

¹⁰ Seventy-three income-support claimants were exempted from participating by a committee due to various personal circumstances, out of a total of 5,700 who were randomized into the treatment during the first sixteen months of the program.

where $Outcome_{ijtp}$ is the outcome of jobseeker *i* assigned to employment office *j*, randomized at time *t*, who belongs to claimant type group *p* (i.e. flow/stock); *Treatment_i* is the indicator for whether jobseeker *i* was assigned to treatment; γ_{jtp} is a fixed effect for the randomization cell (employment office interacted with randomization date and claimant type); X_i is a vector of individual characteristics measured before randomization including age, sex, marital status, number of children, immigration status, education level, indicators for self-reported health limitation, single mother, Ultra-Orthodox Jew, Arab, and vectors for welfare and employment-history indicators in the three years preceding randomization. We cluster standard errors by randomization unit (employment office-randomization month-claimant type), allowing for correlation between the error terms of those who belong to the same pool and office and were randomized at the same time.¹¹

IV. Data Sources

We combine detailed data from various sources to produce a comprehensive picture of each individual before, during, and after the program was implemented. The first administrative data source is the Israeli Employment Service operational database (hereinafter: IES data), which contains basic sociodemographic characteristics of all jobseekers registered with IES, dates of assignment to treatment and control groups, and information on their weekly visits to the employment office. The database includes also the ID number of the jobseeker's spouse as recorded in the Israeli population registry.

The second administrative data source comprises the operational records of the National Insurance Institute of Israel (hereinafter: NII data), which records monthly income-support payments and additional transfer benefits (disability, unemployment, etc.). We combined these data with tax records to determine monthly employment and earnings. The data covers the 2010–2015 period, providing a very comprehensive picture of welfare and labor-market outcomes before, during, and after the intervention for RCT participants and their partners.

¹¹ Abadie et a. (2017) discusses clustering adjustment of standard errors. The authors note that in stratified RCTs where treatment assignment is constant within strata there is no need for adjustment. In our case, we cluster standard errors due to the following reasons. From a sampling design viewpoint, we estimate the program effects using data from a sample of clusters and not the entire population (i.e., we analyzed only data of individuals randomized in the first 12-18 months of the program implementation and from 14 employment offices that participated in the pilot). From an experimental design viewpoint, due to logistical issues, treatment assignment probabilities varied across clusters. An additional justification is provided by Deeb and de Chaisemartin (2021) who show that clustering allows to account for variability in cluster-level shocks that affect the outcome, increasing the external validity of the estimated treatment effects. Overall, our standard errors are smaller without the adjustment but this matters little given that our estimates for the program effects are highly significant.

We complemented these data with survey data that add important insights on the impact of the intervention and the mediating channels. The surveys were administered by IES through a third-party agency in Hebrew and Arabic (for the Jewish and Arab populations, respectively). The first survey took place 12–16 months after the program was launched; the second survey followed the first at a twelve-month interval. The surveys cover a wide range of topics. They include a series of questions that aim to measure non-cognitive skills, work readiness, motivation, and labor-market outcomes such as labor force participation, hours worked, and part-time work that administrative data do not elicit. We provide further details on the survey data in Section VI, where we discuss the mechanisms and additional outcomes.

Sample Construction

The IES operational dataset and the survey data were transferred to the NII Research Department, where they were merged with welfare and tax records hosted at NII using the unique ID number that every Israeli citizen receives upon birth or upon immigration to Israel. The datasets were anonymized and each individual (and spouse, if relevant) was assigned an internal ID number. Given the time frame of the earnings data, we limited the sample to those individuals who were allocated to the treatment or control groups during 2014 in order to be able to follow their labor-market outcomes for twelve months. The analysis sample includes 6,750 individuals. We dropped 599 individuals from the control and treatment groups collectively (about 9% of the sample) who stopped reporting to the employment office before the randomization lists were transferred to the local employment offices.¹² In Appendix Table A1, we show that there is no differential selection of these individuals according to treatment status. This stands to reason because these individuals stopped reporting to IES before knowing their treatment status.

Our final analysis sample includes 6,151 individuals: 3,201 in the control group and 2,950 treated. Table 2 (Column 1) reports the basic demographic characteristics, employment, and welfare history (all included as controls in the analysis of the program effect) of the treatment group as recorded before they were randomized into the program. The table reports balancing tests for each of the individual variables based on regressing each outcome on a treatment dummy and indicators for the randomization block. The table also reports the F-statistic and p-value of a regression that examines whether all covariates can jointly predict treatment status within the randomization cell.

¹² These are individuals whose last visit to the employment office predates their randomization. Compared with the general population of income-support claimants, they are younger, are less likely to report any health limitations, and have a shorter history in the welfare system.

The program participants come from different demographic strands of the Israeli population: 35% Arabs, 19% Ultra-Orthodox Jews, and 21% immigrants. The representation of relatively disadvantaged subgroups is apparent: only 5% have more than twelve years of schooling, 56% have twelve years of schooling, and 39% have fewer than twelve years of schooling. 36.8% report having some health limitation that prevents them from working, 22% are single parents, 52% received income support during the year before randomization, and 24% received income support in the third year before randomization.

There are no systematic differences between the treatment and control groups.¹³ Particularly important is that welfare and employment history of the groups during the three years preceding randomization is balanced. Moreover, the joint significance of all covariates is rejected, suggesting that the ignorability assumption holds, conditional on randomization cell.

V. Results

Program Effects Twelve and Eighteen Months after Randomization

Table 3 (Column 1) reports the effects of the program on the employment, earnings, and welfare outcomes of our main analysis sample as observed twelve months after the randomization date and for outcomes accumulated during the twelve months after randomization. Each cell reports the treatment effect for a specific outcome (along with its standard error) and the respective outcome mean for the control group (in italics). Columns 2 and 3 of the table report similar outcomes for a subset of our main analysis sample that we were able to track for eighteen months given that they were randomized in the first half of 2014.

The results show that the program lowered the probability of reporting to the employment office twelve months after randomization by 15 percentage points (s.e.=0.019)—a significant drop of 38% relative to the outcome mean of the control group (0.384). The program also produced an 8 percentage-point increase (s.e.=0.014) in employment, a 24% upturn in employment relative to the control mean (0.331). Concurrently, the program reduced the likelihood of receiving income support by 11 percentage points (s.e.=0.017) - a 26% decline. The program had no effect on the probability of receiving other NII transfers, such as disability or unemployment compensation. This is important in two different respects. First, it

¹³ We find significant differences in only four out of twenty-five covariates examined. Three differences are significant at the 10% level and only one difference is significant at the 5% level. Moreover, these differences are small in economic terms and are not consistent across covariates.

implies that individuals in the treatment group did not transition to other transfer benefits that might be easier to claim (by not requiring three weekly visits to the employment office, for example). Second, from a fiscal perspective, it means that the savings from the reduction in income-support payments are not offset by other government transfers. Consistent with the increase in employment, we see a significant 12% increase in monthly labor income relative to the control group (161 New Israeli Shekel – NIS in 2016 prices, s.e.=65.48).

Overall, program participants accumulated NIS 2,206 more in labor income twelve months after being assigned to the program than did the control group—a 17% upturn relative to the mean of the control group. Concurrently, they received, on average, NIS 1,860 less in income support (a reduction of 21%). The per-participant cost of the program was NIS 1,400, meaning that the program paid for itself twelve months after an individual is allocated to treatment.

The effects of the program observed twelve months after randomization persisted after eighteen months as well, as seen in columns 2 and 3 of Table 3, which report estimates for a subsample of our main analysis sample to a time horizon of at least eighteen months after randomization. The increase in employment at twelve months among this subsample is of the same order of magnitude as the increase in our main analysis sample and remains similar after eighteen months. This suggests that the increase in employment generated by the program persists at least in the medium term. Concurrently, the positive gap in cumulative earnings between the treatment and the control groups and the negative gap in cumulative income-support payments continued to widen. Thus, the program continues to generate fiscal savings in the longer term.

We also estimate the main effects of the program using individual fixed effects, exploiting our ability to follow individuals before and after randomization into treatment and control groups. We do this by comparing an individual's cumulative income and months employed in the twelve months preceding randomization with the same outcome during the twelve months after randomization, between treated and control individuals. This model can be expressed as follows:

(2)
$$Outcome_{i\tau} = \beta_1 + \beta_2 Treatment_i + \beta_3 post_{\tau} + \beta_4 Treatment_i * post_{\tau} + \delta_i + \varepsilon_{i\tau}$$

where $Outcome_{i\tau}$ is the outcome of jobseeker in period τ (i.e. the year preceding/following the randomization); $Treatment_i$ is the indicator for whether jobseeker *i* was assigned to treatment; $post_{\tau}$ denotes the post-randomization period; and δ_i are individual fixed effects.

The estimates, reported in Table 4, show that the program induced participants to work one additional month (s.e.=0.188) and earn NIS 2,366 (s.e.=916) more than did non-participants during the first twelve months after their being assigned to the program. Compared with the control group, this reflects a 30% increase in employment and a 19% increase in annual labor income. The program led to a decrease of similar magnitude in annual income support (NIS -2,559), leaving total annual income unchanged. These results are reassuring because they strongly resemble the cumulative-outcome estimates reported in Table 3, further supporting the ignorability assumption.

Dynamic Effects of the Program

To get a clearer picture of the impact of the program over time, we examined its effects on a monthly basis. Figure 1a reports the share employed among the treatment and control groups and Figure 1b reports treatment-vs.-control differences in employment along with confidence bands from three years before random allocation to the program to twelve months after that event.¹⁴ The figures show that the treatment and control groups had identical employment trajectories before randomization. Their employment rate was about 32% thirty-six months before randomization. As is typical for populations enrolled in ALMPs, the employment rates of both groups show a decline (the *Ashenfelter dip*) that starts around eighteen months before randomization and accelerates during the year preceding randomization. This is expected because eligibility for the program was based on being unemployed.¹⁵ The employment rates of the treatment and control groups increase over time but the gap between both groups widens month by month. Twelve months after randomization, the control group converges to the employment rate observed three years before randomization (around 33%) while the treatment group surpasses its pre-program employment rate at a record 41%.

The dynamic effects of the treatment also provide interesting insights on how the program works. For example, it is important to examine whether the impacts of the program are driven by the additional requirement that its participants report to IES three times a week instead of one (the threat effect) or by

¹⁴ The means of the treatment group are computed by adding the treatment effect to the outcome means of the control group in order to compare treatment and control groups within the same randomization cell.

¹⁵ Note that the employment rates do not drop to zero at the allocation date because the NII employment records refer to a calendar month while the allocation date may occur at any point during the month. For example, if an individual worked until March 5, 2014, and was assigned to the program on March 20, 2014, she will be recorded as employed on the allocation date. In practice, this creates a slight measurement error for employment spells close to the allocation date, but it matters little for our main results because we focus on medium-term effects. In addition, measurement error in these employment spells should be the same in both the treatment and the control group.

the program's workshops. If the additional requirements push the participants to exit welfare and go to work, we would expect the participants to make an early exit to work, before receiving most of the reemployment services provided by the program, and to show non-existent or negligible exit rates several months into the program.¹⁶ The figure on employment effects suggests that there is an immediate response to treatment in the first two months after assignment to the program but the gaps between the groups widen considerably from month 2 onwards. The treatment effect appears to stabilize around eight months after treatment, consistent with the seven-month maximum duration of the program. The dynamic effects on employment suggest that the program has immediate impacts after enrollment and further impacts after active participation.¹⁷

Figure 2 adds more evidence about the dynamic effects of the intervention by showing treatment and control means and treatment effects on the probability of attending the employment office. By design, all income support claimants attended the employment office by the randomization date. During the first two months after those in the treatment group were assigned to the program, their attendance rate declined by 8 percentage points relative to the control group. Some members of the treatment group transitioned to employment (about 6 percentage points more than the control group) but others (as shown in the next figure) stopped reporting to the employment office despite lacking formal employment. The share of individuals reporting to the employment office continued to decline over time and the gap between the treatment and control group widened until it stabilized at 15 percentage points around eight months after allocation to the program. Roughly, about half of the decline in attendance at the employment office can be attributed to early exits that were probably induced by the additional program attendance requirements while the remaining decline takes place gradually once participants start participating in the workshops.

To complete the picture of the dynamic effects of the program, we plot in Figure 3 the share of individuals who do not attend the employment office, do not receive income support, and do not have formal labor income over time. Here we see that the program induced some individuals (7 percentage points) to stop attending the employment office although they had no formal income (from income support or from work). The gap between the treatment and control group appeared around two or three months after allocation to the program and remained constant thereafter. The drop shortly after allocation suggests

¹⁶ Participants usually start attending workshops one month after being assigned to the program.

¹⁷ An alternative interpretation is that program participants find costlier over time to participate in the workshops and the same job offers become gradually more attractive.

that, for some individuals, the costs associated with the additional program requirement of more intensive attendance to the employment office do not outweigh the benefits of receiving income support. While our data do not allow us to assess this hypothesis formally, it is likely that many of these individuals previously worked in the informal sector and claimed benefits—a behavior no longer available to them once they have to spend several hours per week at the employment office.

Parsing the Average Treatment Effects

The evidence presented above provides important insights on the absence of change, on average, in total income even though the program increases employment and labor income. One possible explanation is that individuals who begin to work lose their eligibility for income support and experience a decline in transfers that fully offsets their gain in income from work. However, Appendix Table A2, plotting the change in monthly income from work, monthly income support transfers, and total monthly income for individuals stratified by their employment and income support status before and after their allocation into the program, shows that this is not the case. In this table, we compare the change in income between twelve months before allocation and twelve months after allocation to the program.¹⁸ We do not intend to claim causality (since we are stratifying by post-treatment outcomes) but to provide a descriptive picture of the income situation of treated individuals twelve months after randomization.

Column 1 of Table A2 reports the change in income of individuals who are formally employed twelve months after their allocation to the program. These individuals experience an increase in income from all sources between the pre- and post-program period. They earn, on average, NIS 2,000 more than what they earned twelve months before allocation to the program and experienced no significant change in income-support transfers, leaving their total income NIS 2,068 higher on average.¹⁹ In contrast, the total income of those who neither work nor receive income support twelve months after allocation to the program falls by NIS 1,216. The last group reported in the table is those who receive income support twelve months after program allocation and do not work: they experience a slight increase in total income (NIS 290) because they gain more from income support than they lose in labor income.

¹⁸ We focus on twelve months before program allocation instead of the months just before allocation in order to avoid a pre-program period that is inherently related to the negative shock that program participants experienced due to becoming eligible.

¹⁹ Note that some employed individuals continue to receive income support in the form of an income supplement (provided their labor income is below a certain threshold).

These descriptive statistics suggest that the zero effect of the program on total reported income hides differential effects among individuals. To parse the average treatment effects, we estimate unconditional quantile treatment effects on total income (from work and from income support) and report them in Figure 4.²⁰ The program does not affect the total income of those at the bottom of the income distribution, who report no income from any source according to the NII records. As noted above, the program induced some individuals to stop reporting to the employment office (and, accordingly, forgoing income support) without obtaining formal employment (an effect of 7 percentage points). As a result, we see a negative treatment effect in the total income of individuals in the 40–50-percentile of total income distribution. Lastly, a positive treatment effect on total income is observed among individuals in income-distribution percentiles 65–75.

Heterogeneous Effects

We also examine heterogeneous treatment effects by individuals' socio-demographic characteristics and pre-program labor-market attachment and welfare dependence. Figure 5 presents the estimated treatment effects on employment for different subgroups along with their confidence band. Sample sizes for each subsample are reported in square brackets. Appendix Table A3 reports estimates of all outcomes for these subsamples. The program increased employment and reduced welfare dependence among almost all groups but had a larger effect (both in absolute terms and relative to the outcome mean of the control group) on some subsamples than on others, e.g., a larger increase in employment among women than among men—8 percentage points (29%) vs. 6 percentage points (16%), respectively. The program was also highly effective among the Arab population, boosting its employment rates by 14 percentage points (an increase of 62%). Positive effects are also observed among the Ultra-Orthodox: the estimate for employment is 0.065 (s.e.=0.044), implying a 16% increase, although the sample is too small to provide a precise estimate. We do observe a positive and significant impact for this population on the number of months worked during the twelve months after allocation to the program: Ultra-Orthodox participants worked, on average, one more month than did non-participants during that time, implying a 29% increase.

²⁰ Given that in our research design treatment status satisfies the independence assumption conditional on the randomization cell, we estimate unconditional quantile treatment effects as developed by Firpo, Fortin, and Lemieux (2009), controlling for randomization cell by applying the algorithm developed by Borgen (2016). Note that this method does not identify the distribution of treatment effects but rather provides estimates for treatment effects on income distribution.

The program is also highly effective among high-school dropouts and those aged thirty-five or older, increasing the employment rate of both groups by 11 percentage points, implying a 40% improvement. Interestingly, the program has a massive impact on those who report health limitations when they register with the employment office, i.e., those who do not receive disability benefits but advise IES upon registration that they have health limitations that impede them from working. Twelve months after randomization, the employment rate of the treated group was 14 percentage points higher than the 24% rate among the control group. The program also raised the monthly income (from work and welfare transfers) of this treated group by NIS 190, i.e., almost NIS 2,000 (11%) in total income accumulated in the twelve months after randomization. The effect of the program on the employment rate of those with no self-reported health limitations was also significant but smaller: 5 percentage points relative to a control mean of 37%.

Two additional groups highly affected by the program are those who have no employment spells in the twenty-four months before randomization into the program and those already on welfare during that period.²¹ The program boosted the employment rate of those in the former group by 9 percentage points (relative to a 17% employment rate in the control group) and of those in the latter group by 11 percentage points (relative to 28% in the control group).

Given the evidence in the literature that ALMPs tend to work best when unemployment is relatively high (Card et al., 2018), we also examine the heterogeneous effects of the program by local unemployment rates. We define low (<7.5%) and high (>=7.5%) unemployment rates relative to the median local unemployment rate (7.5%) across all employment offices participating in the program in 2012, before the program was launched.²² Consistent with previous studies, the effect of the program on participants reporting to offices in high-unemployment areas was larger both in absolute terms and relative to the control mean. Twelve months after randomization, the employment rate of the control group reporting to offices in low-unemployment areas was 42% while that of the control group reporting to offices in high-unemployment areas and a 6 percentage-point (13%) upturn in low-unemployment

²¹ These two groups do not completely overlap. Roughly 40 percent of individuals who have no employment spells during this two-year period receive no income support benefits at the time.

²² The median unemployment rate across all locations of employment offices countrywide is identical to that in the localities of the employment offices analyzed in the sample. The average unemployment rate in Israel during this period (2012) was 6.9%. The interpretation of the results stratified by local unemployment rate should be viewed with caution because we cannot determine whether the larger program impact in high-unemployment areas traces to specific characteristics of welfare claimants, program administrators, or other conditions in these areas.

areas. Similarly, income-support recipiency decreased by 13 percentage points (29%) and 6 percentage points (19%) in high-unemployment and low-unemployment areas respectively.

Household-Level Results

An interesting feature of the program and our data is that we can track and identify the program effect at the household level and look at each partner individually (even if he/she did not participate in the program). Recall that in cases where both partners were eligible, they were jointly assigned to either the treated or the control group. Table 5 reports program effects stratifying the sample by program participation of each partner (both, only one, or single). For comparison purposes, we also report in Column (1) the program effect for the full sample. Overall, we find that the program boosts total household labor income accumulated during twelve months both in households where only one partner was treated and in those where both partners were treated. More interestingly, in two-partner households (columns 2 and 3), the increase in total accumulated household labor income exceeds that in individuals' labor income, implying that the program raises the labor income of both partners. This might be expected among households in which both partners participate in the program (Column 2) but it is an important finding for those households where only one partner is assigned to the program (Column 3).

We further explore this matter by examining the program effects on treated individuals' spouses. We do so by selecting households where only one partner was randomized into treatment or control because the other spouse did not qualify for the program. Focusing on these households, we analyze the spillover effects of the program on the untreated spouse. By virtue of randomization, the spouses of treated and control participants share the same characteristics. In Table 6, we report treatment effects on treated individuals' spouses (Column 1) and separately for men and women (columns 2 and 3 respectively). We find that the program led to an increase in employment and a reduction in welfare dependence among treated women's husbands but induced no change in the labor supply of wives of treated men. This important and novel result suggests that enhancing employment among women in welfare can have positive spillover effects on their husbands' labor supply.

This result lends itself to various possible explanations, such as changes in social norms within the household, information sharing, social networks for employment, and more. Although they cannot be assessed in the context of this study, they provide interesting directions for the design of additional interventions that are part of our future research agenda.

19

Externalities

In addition to its direct effects on its participants, an ALMP has potential indirect effects on nonparticipants. It may affect workers' behavior and options when competing with other participants in the labor market or the firms that employ them. Such externalities make take the form of displacement effects (i.e., program participants taking jobs at non-participants' expense—see, e.g., Blundell et al., 2003; Crépon et al., 2013) or general equilibrium effects through impacts on wages or vacancies (e.g., Gautier et al., 2018). Positive externalities may exist via information sharing or network effects (e.g. Bayer et al., 2008; Hellerstein et al., 2011), peer effects (Manski, 1993) or changes in employment-related social norms (Eugster et al., 2017).

We cannot test each channel individually, but we take a first approach to assess whether there is any evidence on externalities. Similar to the analysis of Crépon et al., (2013), we examine whether the treatment effect is related to the share of income-support claimants assigned to treatment at each employment office in any given month and whether this share affects outcomes of the control group. We discuss the analysis in more detail in Appendix 2 and present the results in Appendix Table A16, where we show that the share treated in a given office at a given month is not related to the probability of reporting to IES for the treatment or the control group.

VI. Assessing the Mechanisms

We now analyze the survey data for additional information on the effect of the program on labor-market outcomes and various measures of non-cognitive skills. These data originate in two separate follow-up surveys conducted by a third party over two periods—February 2015–June 2015 and April 2016– December 2016—capturing individuals fifteen months on average after random assignment.²³ Treated and control groups were contacted by an external company by phone and were told that the survey was meant to produce statistics on individuals who report or reported to IES for the purpose of improving IES customer service. We obtained responses from 2,497 of the 6,151 individuals included in our main analysis

²³ Due to IES logistical constraints, it was not possible to survey each individual at a specific time after randomization. Therefore, the number of months between randomization and the survey date varies across individuals but is balanced across treatment and controls. Individuals in our sample were surveyed between four to thirty-four months after random assignment. The vast majority (86 percent) were polled six months after randomization. The average time was fifteen months and the median ten months.

sample, a 41% response rate.²⁴ Roughly two-thirds of the observations came from the first survey and the rest from the second.²⁵

We examine whether there is differential selection into the survey by treatment status by estimating a linear probability model that estimates the probability of response as a function of personal characteristics, and a treatment dummy controlling for the randomization cell. Results reported in Column 1 of Appendix Table A4, suggest survey response is associated with individuals' characteristics. Namely, the probability of response is higher for individuals with self-reported health limitations, at least twelve years of schooling, income-support recipiency before random assignment, Ultra-Orthodox Jewish identity, and Israeli born. Nevertheless, treatment status is not associated with the probability of responding to the survey. In Column 2, we test for differential selection of treated individuals by personal characteristics by also including interactions between all covariates and the treatment dummy. Only two of the twenty-two treatment indicators are statistically significant. Specifically, we find a negative coefficient only for the interaction of treatment with health limitation and a positive coefficient for the interaction between treatment and Arab indicators. Overall, despite these small imbalances, we do not observe a consistent picture of differential selection into the survey in accordance with treatment status.

To analyze the data yielded by the survey respondents, we construct survey weights to account for nonresponse in order to reflect the characteristics of the entire research population. We estimate a logistic regression model that predicts the likelihood of survey response as a function of treatment assignment, individual characteristics, the interaction between the two, and randomization cell fixed effects (the estimates are reported in Appendix Table A5). Each observation is then weighted by the inverse of the predicted response probability, except for observations of individuals surveyed in both survey waves, which we reweight by half of their assigned weight. In Appendix Table A6, we report the results of a balancing test for the reweighted survey sample, which shows that there are no significant differences between the treatment and comparison groups, both in terms of observable individual characteristics and in the time passed between random assignment to the survey date.²⁶ This table also

²⁴ 567 individuals participated in both surveys.

²⁵ The second survey wave was larger, comprising 1,854 additional individuals who were randomized into treatment and control groups from January 2015 to March 2016. We exclude these observations from the analysis because we wish to focus on the survey sample that coincides with our main sample of individuals who were randomized during 2014, for whom we have complete administrative records on labor-market outcomes and welfare benefits for a duration of at least twelve months after random assignment.

²⁶ There may still be a systematic correlation between unobservables and the propensity to be included in the sample. We cannot entirely rule out this possibility, even though the lack of differences in the observables hints that

shows that the average characteristics of the survey sample are virtually identical to those of the full sample reported in Table 2. Furthermore, we are able to replicate our main results in administrative outcomes obtained for the full sample using the smaller reweighted survey sample (see Appendix Table A7). This is important because it strengths our confidence in using the survey sample to draw conclusions about the effects of the program for the full population.

Survey results

Labor-market outcomes: We begin the survey analysis by exploring the program effects on additional labor-market outcomes that cannot be tested using the administrative data. In particular, we can assess whether the program also affected labor-force participation (by including active job search) and examine hours worked. We estimate the same model as in our main analysis, controlling for survey date. Much as in Table 3, Table 7 displays the program treatment effect on labor-force participation, employment, weekly hours worked, and labor income. The program led to increases of 7.1 and 6.4 percentage points in labor-force participation and employment rates. Thus, it not only boosted employment but also raised the share of individuals who are actively searching for jobs. However, we see no program effect on full-time employment, indicating that the increase in employment rates was driven exclusively by broader incidence of part-time employment. The estimated program effects on the total number of weekly hours and income from work are positive yet only marginally significant. A back-of-the-envelope calculation suggests that the magnitude of these effects almost perfectly corresponds to part-time minimum-wage work by members of the treated group.²⁷

Non-cognitive skills: Having shown that the program improved labor-market outcomes and reduced income-support recipiency, we now examine several channels through which the program may have led to changes in participants' outcomes. We note that we present here evidence on a selected group of channels, because we cannot test every possible mediator. In addition, we cannot manipulate the

the presence of a strong correlation in the unobservables is very unlikely, especially if these unobservables are correlated with the observed covariates.

²⁷ If we assume those who started working because of the program have done so by working in 'half-time' jobs (21.5 hours a week), we would expect an increase of 1.38 hours for the treated group. The estimate we get is just slightly lower (1.24). Similarly, if we assume these jobs are at minimum wage (NIS 23.12 in 2015), and are 'half time' jobs (93 hours a month); we would expect to get an estimated program impact on average monthly income from work of NIS 138 (0.064*23.12*93). This estimate is virtually identical to the estimate we actually get: NIS 141.

channels and assess their effects on outcomes. Nevertheless, we provide important and novel evidence on skills that are affected by the program.

To examine the impact of the program on participants' soft skills, the survey included a series of questions designed to assess individuals' non-cognitive skills and self-perception. These questions were grouped in five modules containing thirty-four items in total. For each individual item, participants were asked to specify the extent to which they agree with various statements on a four or five-point Likert scale (from "strongly agree" to "strongly disagree"). The first module assesses job-search self-efficacy, which refers to individual's confidence in his/her ability to successfully search for a job and perform specific job-search tasks.²⁸ The second module examines work self-efficacy, with which workers' confidence in managing workplace situations such as respecting schedules and collaborating with colleagues is assessed. The third module examines general self-efficacy, which assesses a person's confidence in taking courses of action in a wide array of situations. The fourth module assesses grit: perseverance and passion to achieve long-term goals. The fifth module focuses on self-esteem, which considers individuals' sense of self-efficacy—were included in both survey waves; the grit and self-esteem modules were included only in the second one. This yielded a larger sample size for some of the skills.

The survey questions in each module and their sources are set forth in Appendix 3. To facilitate the interpretation of the results, we reverse the scale of the items so that a higher value denotes a better score and transform each of the items and the aggregate indices into z-scores. In Appendix Table A8, we report the inter-item correlations and Cronbach's Alpha reliability coefficients for the different modules and in Appendix Table A9 we present the correlations among the different aggregate indices. The jobsearch self-efficacy, work self-efficacy, and general self-efficacy domains show high internal consistency (Cronbach's Alpha 0.86, 0.96, and 0.86, respectively) whereas the grit and self-esteem domains have lower levels of consistency (Cronbach's Alpha 0.56 and 0.79, respectively).²⁹

We start by examining the association between these skills and labor-market outcomes using the control group. This is not done to establish causality but to examine the informational content of the survey

²⁸ Job search self-efficacy can be affected by learned skills and self-perception.

²⁹ We obtain very similar results based on McDonald's omega (McDonald, 1999): job-search self-efficacy=0.864, work self-efficacy=0.963, general self-efficacy=0.863, grit=0.491, self-esteem=0.776.

indices. ³⁰ For this purpose, we regress each of the survey labor-market outcomes (labor-force participation, employment, full-time employment, weekly hours worked, and labor earnings) on the mean standardized scores of each of the five modules while controlling for individual characteristics. The results (Table 8) show that all skills are positively correlated with better labor-market outcomes.

We then examine the effect of the program on these skills by plotting the cumulative distributions (CDFs) of these skills for the treatment and control groups along p-values for Mann-Whitney tests of stochastic dominance (Figure 8).³¹ The CDFs of the treatment group for job-search efficacy, work-self-efficacy, and self-esteem are shifted to the right relative to those of the comparison group, suggesting that the program indeed improved these skills. This is also confirmed by p-values of Mann-Whitney tests that reject the null hypothesis for equality of distributions between the treated and control groups. In contrast, no significant differences emerge between the CDFs of the treatment and control groups for grit or general self-efficacy.

Reported next are regression coefficients of average treatment effects for each category, based on a system of seemingly unrelated regressions that treat the items in each category as a family of outcomes (Table 9). This method takes into account that the outcomes in each category are correlated by allowing for individual-level correlation of the error terms across equations (e.g., Kling et al., 2007).³² The effects on each individual item are presented in Appendix Tables A10-A14.

The results suggest a significant and positive effect of the program on its participants' self-reported jobsearch efficacy and work self-efficacy, which increased by 6% and 9% of a standard deviation relative to the control group. We also find a 6% of a standard deviation increase in self-esteem, although the standard errors are too large (due to the smaller sample) to reject the null hypothesis of zero effect on this domain. Consistent with the evidence presented in Figure 8, we find no effect on general self-efficacy and on grit. Overall, these findings suggest that the program enhances its participants' work-related skills.

³⁰ Conducting an equivalent exercise using the available administrative labor-market outcomes, we found a similar pattern (results not shown).

³¹ To compare the distributions, we use residualized z-scores that we obtain by regressing each z-score on the vector of individual's characteristics. To account for the randomization block fixed effects, we apply inverse probability weighting, weighting treated observations by 1/p and control observations by 1/(1-p) (where p is the proportion treated in the randomization block). We then adjust the weights for those surveyed twice by dividing by two, trim weights to the 90th percentile to avoid extreme values, and normalize them to make sure they add up to 1 for each group and reflect the total sample size.

³² That is, we define the average treatment effect for category c as $\tau_c = \frac{1}{K_c} \sum_{k=1}^{K_c} \frac{\pi_{kc}}{\sigma_{kc}}$ where K_c is the number of outcomes included in category c, π_{kc} is the effect on outcome k included in category c, and σ_{kc} is the standard deviation of the outcome. We treat (σ_{kc}) as known based on the results of Kling and Liebman (2004) and given that we have a large sample.

The lack of an effect on grit is consistent with evidence from recent interventions among youths in Chicago that did not lead to an improvement in grit (Heller et al., 2017) and a recent meta-analytic review of the grit literature (Credé et al., 2017). The contrasting results that we obtain when assessing the effects on work self-efficacy versus general self-efficacy are consistent with Bandura (1986), the forefather of the self-efficacy concept, who claims that self-perceptions of efficacy vary across activity domains and recommends testing self-efficacy measures tailored to a particular domain rather than to a global disposition.

Overall, the analysis of non-cognitive skills shows that that the program does affect outcomes related to its core objectives. The fact that we do not observe effects on skills that were not the core objective of the program suggests that the positive effects observed for the relevant outcomes are not driven by an attempt by program participants to please the survey interviewer and do not result from possible biases that result from survey-respondent selection. ³³ It is tempting to perform a back-of-the-envelope calculation to assess whether the increase in non-cognitive skills can explain the increase in employment using the observed correlations between non-cognitive skills and employment for the control group and the estimated treatment effect. We are agnostic about such calculations because we do not know how these skills complement each other and which other skills may have improved. Instead, we see the positive effects on these skills as evidence of the possibility of affecting these skills among adults and as strong validation of the finding that the program tools worked in the expected direction.

VII. Conclusions

A vast literature stresses the importance of non-cognitive skills for human capital formation and labor market success. Yet there is little evidence about the returns to investments in these skills, especially among adults. This study examines the impact of an active labor-market program implemented in Israel that focuses on enhancing welfare recipients' non-cognitive skills in order to prepare them for successful

³³ An additional possible explanation for the lack of an effect in some domains is lack of power. For example, a two independent means power analysis (with error probabilities $\alpha = 0.05$ and $\beta = 0.8$ and the relevant proportion of treated and control), using the means of the residuals of these scores after controlling for observables and randomization cell, suggests that to detect a difference of 6% or 9% of a standard deviation in grit, we would need a sample of roughly 7,235 treated observations and 3,217 controls. Another possible explanation for the lack of an impact on grit is the lower degree of reliability of this domain. Although this seems less likely because despite its lower reliability score, we still found that it was strongly associated with labor-market outcomes among the control group.

immersion in the labor market. Using a randomized-control trial, we estimate the effect of the program on a wide range of outcomes and examine the mechanisms through which the program works.

The results point to positive and significant effects on labor-force participation, employment rates, and labor income. The effects on hours worked and wages suggest that those who enter the labor market do so in part-time jobs that pay close to minimum wage. We also find a significant negative effect on incomesupport recipiency and, correspondingly, on the size of income-support payments received by those assigned to the program with no evidence of substitution with alternative benefits (e.g., disability). The cost of the program per participant is more than outweighed by savings on government welfare transfers within twelve months. The program has a stronger impact on individuals who have lower labor-force attachment and longer history in the welfare system, fewer than twelve years of schooling, and self-reported health limitations. Interestingly, it also increased employment among treated women's husbands without doing so among treated men's wives. We also find no evidence of spillover effects among the control group.

Overall, the program reduced the share of treated individuals who report to the employment office. The total decrease may be decomposed into two separate channels that affect different individuals. Half of the effect is driven by individuals who stop reporting to the employment office due to the additional program requirements. These individuals forgo income-support transfers and are not found to be employed in the formal sector, whereas it is likely that they are employed in the informal sector. The other half stop reporting to the employment office because they have become employed.

An analysis of the survey data shows that treated individuals score higher in measures of work self-efficacy and job-search-efficacy, and marginally, on self-esteem, relative to the control group. These non-cognitive skills are associated with superior labor-market outcomes and, as such, may mediate the impact of the program on employment. Our study shows that it is possible to enhance the work-related attitudes and self-perception of long-term unemployed individuals in a cost-effective way, leading to an increase in their employment and earnings. These effects have also positive spillovers within households, making such programs all the more attractive.

26

References

- Abadie, A., Athey, S., Imbens, G. W., and Wooldridge, J. "When should you adjust standard errors for clustering?" Working Paper 24003, National Bureau of Economic Research (2017).
- Bandura, Albert. "The explanatory and predictive scope of self-efficacy theory." *Journal of Social and Clinical Psychology* 4.3 (1986): 359-373.
- Bayer, Patrick, Stephen L. Ross, and Giorgio Topa. "Place of work and place of residence: Informal hiring networks and labor market outcomes." *Journal of Political Economy* 116.6 (2008): 1150-1196.
- Black, Dan A., Jeffrey A. Smith, Mark C. Berger, and Brett J. Noel. "Is the threat of reemployment services more effective than the services themselves? Evidence from random assignment in the UI system." *American Economic Review* (2003): 1313-1327.
- Blattman, Christopher, Julian C. Jamison, and Margaret Sheridan. "Reducing crime and violence: Experimental evidence from cognitive behavioral therapy in Liberia." *American Economic Review* 107.4 (2017): 1165-1206.
- Blundell, Richard, M. Costa Dias, and Costas Meghir. "The impact of wage subsidies: a general equilibrium approach." Institute of Fiscal Studies and Bank of Portugal (2003).
- Brunello, Giorgio, and Martin Schlotter. "Non-cognitive skills and personality traits: Labour market relevance and their development in education & training systems." IZA Discussion Paper No. 5743. (2011)
- Borgen, Nicolai T. "Fixed effects in unconditional quantile regression." *The Stata Journal* 16.2 (2016): 403-415.
- Card, David, Jochen Kluve, and Andrea Weber. "What works? A meta analysis of recent active labor market program evaluations." *Journal of the European Economic Association* 16.3 (2018): 894-931.
- Carneiro, Pedro Manuel, and James J. Heckman. "Human capital policy." (2003).
- Credé, Marcus, Michael C. Tynan, and Peter D. Harms. "Much ado about grit: A meta-analytic synthesis of the grit literature." *Journal of Personality and Social Psychology* 113.3 (2017): 492.
- Crépon, Bruno, Esther Duflo, Marc Gurgand, Roland Rathelot, and Philippe Zamora. "Do labor market policies have displacement effects? Evidence from a clustered randomized experiment." *The Quarterly Journal of Economics* 128, no. 2 (2013): 531-580.
- Crépon, Bruno, and Gerard J. Van Den Berg. "Active labor market policies." *Annual Review of Economics* 8 (2016): 521-546.
- Deeb, A. and de Chaisemartin, C. Clustering and external validity in randomized controlled trials. Working Paper (2020).

- Dolton, Peter, and Donal ONeill. "The long-run effects of unemployment monitoring and work-search programs: experimental evidence from the United Kingdom." *Journal of Labor Economics* 20, no. 2 (2002): 381-403.
- Eugster, Beatrix, Rafael Lalive, Andreas Steinhauer, and Josef Zweimüller. "Culture, work attitudes, and job search: Evidence from the Swiss language border." *Journal of the European Economic Association* 15.5 (2017): 1056-1100.
- Firpo, Sergio, Nicole M. Fortin, and Thomas Lemieux. "Unconditional quantile regressions." *Econometrica* 77.3 (2009): 953-973.
- Gautier, Pieter, Paul Muller, Bas van der Klaauw, Michael Rosholm, and Michael Svarer. "Estimating equilibrium effects of job search assistance." *Journal of Labor Economics* 36.4 (2018): 1073-1125.
- Greenberg, David H., Charles Michalopoulos, and Philip K. Robins. "A meta-analysis of governmentsponsored training programs." *Industrial & Labor Relations Review* 57, no. 1 (2003): 31-53.
- Greenberg, David H., Andreas Cebulla, and Stacey Bouchet. *Report on a meta-analysis of welfare-to-work programs*. Institute for Research on Poverty, 2005.
- Heckman, James J., Jora Stixrud, and Sergio Urzua. "The effects of cognitive and noncognitive abilities on labor market outcomes and social behavior." *Journal of Labor Economics* 24.3 (2006): 411-482.
- Heckman, James J., and Tim Kautz. "Hard evidence on soft skills." *Labour Economics* 19, no. 4 (2012): 451-464.
- Heller, Sara B., Anuj K. Shah, Jonathan Guryan, Jens Ludwig, Sendhil Mullainathan, and Harold A. Pollack.
 "Thinking, fast and slow? Some field experiments to reduce crime and dropout in Chicago." *The Quarterly Journal of Economics* 132.1 (2017): 1-54.
- Hellerstein, Judith K., Melissa McInerney, and David Neumark. "Neighbors and coworkers: The importance of residential labor market networks." *Journal of Labor Economics* 29.4 (2011): 659-695.
- Kling, Jeffrey R., Jeffrey B. Liebman, and Lawrence F. Katz. "Experimental analysis of neighborhood effects." *Econometrica* 75.1 (2007): 83-119.
- Kautz, Tim, James J. Heckman, Ron Diris, Bas Ter Weel, and Lex Borghans. Fostering and measuring skills: Improving cognitive and non-cognitive skills to promote lifetime success. No. w20749. National Bureau of Economic Research, 2014.
- Kluve, Jochen. "*The effectiveness of European active labor market programs.*" *Labour Economics* 17, no. 6 (2010): 904-918.
- Kugler, Adriana, Maurice Kugler, Juan Saavedra and Luis Omar Herrera Prada, (forthcoming). "Long-Term Educational Consequences of Vocational Training in Colombia: Impacts on Young Trainees and their Relatives". *The Journal of Human Resources*.

- Lindqvist, Erik, and Roine Vestman. "The labor market returns to cognitive and noncognitive ability: Evidence from the Swedish enlistment." *American Economic Journal: Applied Economics* 3.1 (2011): 101-28.
- Manski, Charles F. "Identification of endogenous social effects: The reflection problem." *The Review of Economic Studies* 60.3 (1993): 531-542.
- McDonald, R. P. "Test theory: a unified treatment." Lawrence Earlbaum Associates. Inc., Mahwah, NJ (1999).





Notes: The left panel plots employment rates for the treated and the control group by months since randomization. The right panel plots the program effect on employment with a 95 percent confidence interval. Month zero corresponds to month of random assignment.



Figure 2: Dynamic effects - share reporting to employment office

(a) Levels

(b) Treatment-Control

Notes: The figure reports the program effect on reporting to the employment office with a 95 percent confidence interval, and the difference in reporting rates between the treated and control groups, over time. Month zero corresponds the month of random assignment.



Figure 3: Dynamic effects - share not employed, not reporting to employment office and not recieving income support

Notes: The figure reports the program effect on the probability not to be reporting to the employment office while not working nor receiving income support benefits with a 95 percent confidence interval, and the difference in this share of individuals between the treated and control groups, over time. Month zero corresponds the month of random assignment.



Figure 4: Quantile treatment effects on the distribution of total income

Notes: The figure reports the program effect for each percentile of the total income (i.e labor earnings and income support) distribution 12 months after random assignment with a 95 percent confidence interval.



Figure 5: Heterogeneous Employment Effects of the Program

Notes: The figure reports the program's impact on employment across different subpopulations with 95 percent confidence intervals. Number of observations are reported in brackets.



Figure 6: Local labor market treatment intensity across individuals

Notes: The figure reports the distribution of the local labor market treatment intensity among individuals in our sample according to their employment office and month of assignment

Figure 7: Residual variance of labor market treatment intensity



Notes: The figure reports the residual variation in local labor market treatment intensity when controlling for employment office and month fixed effects



Figure 8: Non-cognitive skills distributions

(e) Grit

Notes: These figures plot the cumulative distribution functions of each of the non-cognitive scores by treatment status. Reported p-values refer to the results of the Mann Whitney Wilcoxon test for difference in distributions between treatment and control. 5

	Employment offices in	All other employment offices	
	the RCT		
	(1)	(2)	
Number of active job-seekers	25,459	30,973	
Age	38.2	38.4	
Education	9.3	9.6	
Number of supported children	2.8	2.4	
Women	0.61	0.64	
Married	0.52	0.47	
Arab	0.64	0.54	
Immigrant	0.13	0.16	
Locality S.E.S	5.0	5.1	
Local unemployment rate	0.065	0.072	
Ν	14	57	

Table 1. Employment Offices in the Experiment versus All other Offices

Notes: The table reports the population characteristics and local labor market conditions in employment offices included in the RCT and in the remaining employment offices in Israel. The number of job seekers and their average characteristics are based on all active income support claimants aged 18-50 in the IES system in March 2014. The local unemployment rate is the population-weighted average of localities in the catchment area of the employment offices in each group. Locality S.E.S is the population-weighted average S.E.S index of localities in the catchment area of the employment offices in each group in 2012. The S.E.S index is published by The Central Bureau of Statistics (CBS) and ranges from 1 (lower SES) to 10 (highest SES).

Table 2. Descriptive Statistics and Balancing Tests								
	Treated (1)	T-C (2)		Treated (1)	T-C (2)			
Female	0.544	-0.011 (0.018)	Months worked months [-12;0]	2.82	0.003 (0.129)			
Age	34.57	0.169 (0.263)	Months worked months [-24;-11]	3.93	0.068 (0.141)			
Married	0.473	0.004 (0.012)	Months worked months [-36;-23]	4.29	0.143 (0.149)			
Children	2.00	0.061 (0.068)	Total earnings months [-12;0]	9754	80 (614)			
Single parent	0.219	0.003 (0.012)	Total earnings months [-24;-11]	16320	680 (820)			
Immigrant	0.208	-0.024* (0.013)	Total earnings months [-36;-23]	18242	860 (871)			
Self-reported health limitation	0.362	0 (0.013)	Total income support months [-12;0]	5946	250 (326)			
Arab	0.347	0.011 (0.011)	Total income support months [-24;-11]	3755	220 (269)			
Ultra Orthodox	0.189	0.019** (0.009)	Total income support months [-36;-23]	3211	190 (208)			
Less than 12 years of schooling	0.394	-0.028* (0.015)	Months since registration	3.36	-0.056 (0.000)			
12 years of schooling	0.555	0.029* (0.016)	F-Stat for joint significance	1.01				
More than 12 years of schooling	0.050	0 (0.008)	P-value Number of observations	0.45 3201	6151			
Received income support months [-12;0]	0.523	0.013 (0.013)						
Received income support months [-24;-11]	0.270	0.004 (0.016)						
Received income support months [-36;-23]	0.236	0.007 (0.013)						

Table 2. Descriptive Statistics and Balancing Tests

Notes: The table reports the average characteristics of treatment group participants (column 1) alongside the estimated difference with the control group conditional on randomization unit fixed effects (column 2). The reported F statistic tests the joint significance of all covariants in a linear probability model predicting treatment status conditional on randomization unit fixed effects. Monetary values in real 2016 NIS. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.
	12 months horizon		
	sample	18 months hori	
	Impact after 12		Impact after 18
	months	Impact after 12 months	months
	(1)	(2)	(3)
Reporting to employment office	-0.15***	-0.171***	-0.133***
	(0.019)	(0.027)	(0.027)
	0.384	0.405	0.330
Employed	0.079***	0.089***	0.082***
	(0.014)	(0.022)	(0.025)
	0.331	0.326	0.353
Income from work	161**	200*	276**
(Including zeroes)	(65)	(114)	(121)
	1,345	1,341	1,422
Cumulative income from work	2026***	2130**	3334**
(Including zeroes)	(563)	(902)	(1404)
	12,301	11,897	20,306
Received Income support	-0.105***	-0.132***	-0.105***
	(0.017)	(0.024)	(0.022)
	0.408	0.415	0.360
Income support payments	-170***	-233***	-184***
(Including zeroes)	(29)	(41)	(41)
	625	651	562
Cumulative income support	-1860***	-2300***	-3507***
(Including zeroes)	(278)	(376)	(558)
	8,813	8,994	12,576
Total Income	-9	-33	92
(Including zeroes)	(72)	(108)	(119)
	1,971	1,992	1,984
Total cumulative income	167	-171	-173
(Including zeroes)	(663)	(908)	(1372)
	21,114	20,891	32,881
Received other welfare payments	-0.009	-0.002	-0.01
(disability or UI or other)	(0.009)	(0.017)	(0.019)
	0.111	0.112	0.134
Ν	6151	1643	1643

Notes: The table reports the program effect on participants' outcomes. Controls include sex, marital status, age, number of children, schooling level, indicators for new immigrant, single mothers, Arab, ultra-orthodox Jew, self-reported health limitations, vectors for employment, income from work and welfare history, and randomization unit fixed effects. Monetary values in real 2016 NIS. Control group means in italics. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

		Cumulative	Cumulative	Total
	Total months	income from	income	cumulative
	employed	work	support	income
	(1)	(2)	(3)	(4)
Post	0.53***	2716***	3711***	6427***
	(0.533)	(756)	(299)	(780)
Treatment * Post	1.003***	2366***	-2591***	-224
	(0.188)	(912)	(386)	(969)
Constant	2.783***	9673***	5541***	15214***
	(0.057)	(261)	(136)	(268)
N	12,302	12,302	12,302	12,302

Table 4. Program Effects from Individual Fixed Effects Model: 12 Months After Randomization - 12 Months Before Randomization

Notes: The table reports the program effect on participants' cumulative outcomes while controlling for individual fixed effects. The sample includes two observations per individual: one measurement for cumulative outcomes for the year that preceded randomization and the second measurement for cumulative outcomes for the twelve months post-randomization. Monetary values in real 2016 NIS. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

Table 5. Program Ef		Both spouses	Only one spouse	
	All	assigned	assigned	Singles
	(1)	(2)	(3)	(4)
Reporting to employment office	-0.15***	-0.233***	-0.14***	-0.133***
	(0.019)	(0.048)	(0.031)	(0.019)
	0.384	0.526	0.350	0.349
Employment	0.079***	0.109***	0.078***	0.075***
	(0.014)	(0.036)	(0.023)	(0.021)
	0.331	0.231	0.308	0.382
ncome from work	161**	300*	57	192*
Including zeroes)	(65)	(159)	(115)	(99)
	1,345	0,841	1,309	1,532
cumulative income from work	2026***	2407**	2258**	1811**
Including zeroes)	(563)	(1194)	(1040)	(851)
	12,301	7,566	11,617	14,324
Received Income support	-0.105***	-0.236***	-0.095***	-0.073***
	(0.017)	(0.060)	(0.028)	(0.020)
	0.408	0.630	0.389	0.347
ncome support payments	-170***	-324***	-160***	-147***
ncluding zeroes)	(29)	(79)	(40)	(43)
	625	809	552	615
umulative income support	-1860***	-3140***	-1838***	-1624***
including zeroes)	(278)	(699)	(503)	(412)
	8,813	10,583	8,004	8,786
otal Income	-8.9	-24.8	-102.1	45.4
Including zeroes)	(71.6)	(149.6)	(119.3)	(108.2)
	1,971	1,650	1,860	2,147
otal cumulative income	167	-734	420	187
Including zeroes)	(663)	(1197)	(1205)	(1002)
	21,114	18,149	19,622	23,110
Received other welfare payments (disability or UI or	-0.009	0.006	0.007	-0.02
other)	(0.009)	(0.014)	(0.016)	(0.014)
	0.111	0.048	0.072	0.152
IH level - Income from work	283***	647*	324	192*
Including zeroes)	(102)	(343)	(227)	(99)
	2,114	1,746	3,270	1,532
IH level - cumulative Income from work	3399***	6827**	4574**	1811**
Including zeroes)	(893)	(2716)	(2140)	(851)
	20,213	15,747	32,505	14,324
IH level - Income support payments	-257***	-664***	-255***	-147***
Including zeroes)	(40)	(155)	(70)	(43)
	0,900	1,617	0,967	0,615
IH level - Cumulative income support	-2844***	-6186***	-3274***	-1624***
Including zeroes)	(363)	(1300)	(811)	(412)
U ,	12,596	21,240	13,991	8,786
IH level - Total Income	26	-17	69	45
Including zeroes)	(101)	(313)	(216)	(108)
	3,014	3,363	4,237	2,147
IH level - Total cumulative income	555	641	1301	187
Including zeroes)	(915)	(2584)	(2088)	(1002)
	32,809	36,986	46,496	23,110
	0_,000	1045	,	3259

Notes: The table reports the program effect on individual and household level outcomes by program participation status of each of the partners. Column (1) reproduces the main results reported in column (1) of table 3. Column 2 reports treatment effects for individuals from households where both partners were allocated to the program. Column 3 reports treatment effects for individuals from households where only one partner was allocated to the program. Column 4 reports treatment effects for individuals from single-headed households. All regressions control for the same set of covariates reported in Table 3 and include randomization unit fixed effects. Monetary values in real 2016 NIS. Control group means in italics. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

	Effect by Spouses Treat Spouse assigned to	Wife assigned to	Husband assigned to
	treatment	treatment	treatment
	(1)	(2)	(3)
Reporting to employment office	-0.14***	-0.174***	-0.089**
	(0.030)	(0.039)	(0.038)
	0.350	0.385	0.304
Employment	0.039	0.058*	0.005
	(0.029)	(0.033)	(0.040)
	0.400	0.475	0.302
ncome from work	266	458	-150
Including zeroes)	(187)	(292)	(204)
	1,961	2,546	1,195
Cumulative income from work	2316	4473	-1195
Including zeroes)	(1782)	(2799)	(2095)
	20,888	26,768	13,187
Received Income support	-0.063**	-0.115***	0.03
	(0.025)	(0.033)	(0.039)
	0.331	0.359	0.294
ncome support payments	-95***	-151***	4
Including zeroes)	(35)	(47)	(55)
-	415	451	369
Cumulative income support	-1436***	-1828***	-536
Including zeroes)	(401)	(523)	(551)
	5,986	6,311	5,562
otal Income	170.9	307.2	-146.2
Including zeroes)	(185.1)	(293.7)	(195.4)
	2,377	2,997	1,564
Fotal cumulative income	880	2645	-1731
Including zeroes)	(1772)	(2761)	(1920)
	26,874	33,079	18,749
Received other welfare payments (disability or	0.008	0.012	-0.013
JI or other)	(0.017)	(0.023)	(0.025)
	0.090	0.090	0.090
HH level - Income from work	324	554*	-100
Including zeroes)	(217)	(308)	(313)
	3,270	3,224	3,329
H level - cumulative Income from work	4574**	5885*	1375
Including zeroes)	(2047)	(3005)	(3008)
5 ,	32,505	33,020	31,831
H level - Income support payments	-255***	-410***	-59
Including zeroes)	(67)	(72)	(95)
	0,967	1,133	0,749
H level - Cumulative income support	-3274***	-4225***	-1664
Including zeroes)	(775)	(846)	(1056)
	13,991	15,870	11,530
		,	
HH level - Total Income	69	144	-159
Including zeroes)	(206)	(306)	(289)
	4,237	4,358	4,079
H level - Total cumulative income	1301	1660	-289
Including zeroes)	(1997)	(2880)	(2747)
	46,496	48,890	43,361
d	1845	1013	832

Notes: The table reports the program effect on the individual and household level outcomes for non-treated spouses of program participants. The control group for this sample includes spouses of individuals assigned to the control group. Column (1) reports estimates for spouses of both genders and columns (2) and (3) reports estimates for men and women respectively. All regressions control for the same set of covariates reported in Table 3 and include randomization unit fixed effects. Monetary values in real 2016 NIS. Control group means in italics. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

	Treatment effect (1)
LFP	0.071***
	(0.018)
	0.562
Employment	0.064***
	(0.023)
	0.344
Full time employment	0.01
	(0.015)
	0.170
Hours worked	1.244*
(zero for the unemployed)	(0.730)
	10.009
Monthly income from work	141
(zero for the unemployed)	(90)
	1164
Number of observations	3,064

Table 7. Program Effect on Labor Market Outcomes from Survey

Notes: The table reports the program effect on participants' self-reported labor market outcomes among the survey sample. All regressions control for the same set of covariates reported in Table 3 and include randomization unit fixed effects. Observations are weighted by survey weights. Monetary values in real 2016 NIS. Control group means in italics. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

	Job search self efficacy score (1)	Work self efficacy score (2)	Self efficacy score (3)	Self esteem score (4)	Grit score (5)
Labor Force Paricipation	0.17***	0.128***	0.065***	0.078***	0.109***
	(0.015)	(0.013)	(0.015)	(0.026)	(0.025)
	<i>0.562</i>	<i>0.562</i>	<i>0.562</i>	<i>0.562</i>	<i>0.562</i>
Employment	0.065***	0.065***	0.034**	0.049*	0.065***
	(0.016)	(0.011)	(0.015)	(0.026)	(0.022)
	<i>0.344</i>	<i>0.344</i>	<i>0.344</i>	<i>0.344</i>	<i>0.344</i>
Full time employment	0.038***	0.031***	0.032**	0.057**	0.053***
	(0.013)	(0.009)	(0.013)	(0.024)	(0.017)
	<i>0.170</i>	<i>0.170</i>	<i>0.170</i>	<i>0.170</i>	<i>0.170</i>
Hours worked (zero for the unemployed)	2.235*** (0.525) <i>10.009</i>	1.845*** (0.399) <i>10.009</i>	1.245** (0.581) <i>10.009</i>	2.978*** (1.070) <i>10.009</i>	3.037*** (0.852) <i>10.009</i>
Monthly income from work (zero for the unemployed)	262.431*** (70.702) <i>1164.280</i>	210.214*** (54.129) <i>1164.280</i>	148.454** (71.949) <i>1164.280</i>	245.482* (147.882) 1164.280	289.712** (115.286) <i>1164.280</i>

Table 8. Association Between Non-Cognitive Skills and Labor Market Outcomes Based on the Control Sample

Notes: The table reports the association between standardized aggregate non-cognitive scores and self-reported labor market outcomes among the control group. Each cell reports estimates from a separate regression. Controls include sex, marital status, age, number of children, schooling level, indicators for new immigrant, single mothers, Arab, ultra-orthodox Jew, self-reported health limitations, vectors for employment, income from work and welfare history. Observations are weighted by survey weights. Monetary values in real 2016 NIS. Labor market outcomes means in italics. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

Table 9. Program Effect on Non-Cognitive Skills				
	Ν	Treatment effect		
	(1)	(2)		
Job search self efficacy score	2701	0.059*		
		(0.035)		
Work self efficacy score	2711	0.085**		
		(0.039)		
Self efficacy score	2753	0.005		
		(0.042)		
Grit score	831	-0.023		
		(0.042)		
Self esteem score	853	0.059		
		(0.049)		

Notes: The table reports the program effect on participants' non-cognitive skills based on a set of seemingly unrelated regressions for each group. Estimates for the individual items are reported in Tables A11-A15. All regressions control for the same set of covariates reported in Table 3 and include also survey month and randomization unit fixed effects. Observations are weighted by survey weights. Standard errors clustered at the randomization unit level in parentheses. * p < 0.05, ** p < 0.01, *** p < 0.001.

Appendix

Appendix 1: Program Details

Employment Circles is an Active Labor Market Program that aims to re-integrate chronically unemployed income-support claimants into the labor force by providing them with personalized treatment composed of various occupational workshops.

After being assigned to the program, participants start with an initial intake meeting where their ability and motivation to go back to work are assessed. Intake comprises two individual meetings with an occupational trainer who diagnoses the participant in terms of employability, level of motivation, and barriers to employment, and makes a recommendation for a specific program track based on this diagnoses. A final decision is then made by the head of the employment office and the caseworker for program participants in the office. The personalized program track is composed of weekly meetings with the caseworker and a combination of some or all of the following four workshops:

• Purpose-focused preparatory workshop

Designed to prepare relatively low and medium motivated individuals for the job-search phase or the personal-skills workshop, focusing on improving job-search motivation and boost self-esteem and self-efficacy. Main objectives: increase participant's motivation, identify his/her strengths, and foster his/her career self-image and belief in work capacity. Consists of group sessions and personal meetings, four hours a week (two hours twice a week) for three weeks. Group sessions are devoted to identifying each participant's strengths and skills, familiarizing him/her with different types of work environments, and setting career aspirations and employment goals.

• Job-placement-focused preparatory workshop

Designed to prepare medium or relatively high motivated individuals and those who finished the *Purpose-focused preparatory workshop* for the job-search phase while focusing on providing job-seeking skills. Consists of group sessions and two-hour personal meetings held twice a week for three weeks. Content includes fostering self-introduction skills, acquiring job-search skills with emphasis on entry-level jobs, writing a résumé, and job-interview and assessment-center simulations. At the meetings, each participant defines a set of entry-level jobs and builds a program to achieve the job search goals.

Personal-skills workshop

An intensive workshop designed to build a career path and foster self-motivation, work selfefficacy, and interpersonal skills of program participants with low-to-medium job readiness. Consists of group sessions and personal meetings, ten hours per week (five hours twice per week) for 6 weeks. Content includes vocational guidance, positive self-talk, conflict resolution, dealing with personal obstacles and new tasks, better handling of feedback, and fostering excellence on the job. The workshop puts a special emphasis on the group dynamics in order to build social support and push participants to progress together as a group.

• Job-search workshop and group coaching

Supervised pro-active job search in a computer lab, four hours per week for up to four months. Participants are encouraged to search for suitable entry-level jobs that match their capabilities and aspirations and have a future growth trajectory, all in accordance with the participant's personal job-search program and goals. The meetings include both group and individualized coaching to provide feedback and group support in the job-search process.

Program participants must report to the labor office three times per week: twice for workshop participation and once for an individual meeting with their caseworker. The workshops are conducted by qualified occupational trainers and coaches that provide each participant with the personal attention needed to identify and remove the obstacles that stand in the way of his/her success in the workplace.

Appendix 2. Externalities

To examine the program externalities, we test whether the share of income support claimants assigned to the program in a given office and month is associated with outcomes for the treated or the control group. We have information only on treated and control individuals, so we cannot assess the effects on individuals outside this sample. Still, we think that given the focus of the program on individuals who receive welfare benefits, the most relevant group that may be affected are other welfare recipients because they have similar skills, earnings, and employment potential. In addition, given the small size of the treated population relative to the size of the labor market, we assume that the likelihood of general equilibrium effects of the program on the labor market even at the local level is rather low.

For this analysis, we expand the sample to include jobseekers who were randomized into the program between January 2015 and February 2016 and focus on the effect of the program on the probability of reporting to IES twelve months after randomization. We select this larger sample in order to obtain greater variation in the proportion of treated individuals within employment offices over time and to increase power (increasing the chances of detecting externalities in case they exist). The sample expansion leads us to focus on the probability of reporting to IES as the main outcome of interest because data on this outcome are available to us over a longer time horizon (as opposed to employment and welfare transfers, which are available only up to 2015).¹ We restrict the sample to jobseekers who were randomized from the incoming flow of claimants and define the fraction of job seekers assigned to treatment as the share of treated individuals in the monthly incoming flow of claimants at each employment office.² The share of monthly treated individuals varies considerably across employment offices and over time due to regular fluctuations in the incoming flow of claimants and the capacity of the program at the employment office. Figure 6 presents the overall distribution of the share of treated individuals across offices and time. A variance decomposition analysis indicates that within-office variation accounts for nearly 80% of total variation. The residual variation in the monthly share of treated individuals, controlling for employment office and month fixed effects is shown in figure 7. This is the

¹ Any effect on employment is expected also to be reflected in the probability of reporting to IES. Thus, the absence of an effect on the probability of reporting to IES is a good indicator of the lack of an effect on employment.

² In principle, we could have focused on the share of treated individuals in the same locality of residence rather than the locality of the employment office attended. However, given that many job seekers reside in relatively small localities and that the catchment areas of employment offices largely overlap with local labor markets, we prefer to focus on the latter definition. In addition, we defined the share treated based on the monthly incoming flow of welfare claimants because it is clearly defined unlike the share treated among the welfare stock. Our results are robust to alternatives that include the incoming flow of UI claimants in the denominator (results not shown).

variation exploited in the analysis. We show in Appendix Table A15 that within office fluctuations in the share of treated individuals are not related to jobseekers' characteristics either overall or specifically among members of the treated or the control group.

To assess the possibility of program externalities, we estimate the following equation:

(3) IES_attendance_{ijt} = $\beta_0 + \beta_1$ Treatment_i + β_2 Share_treated_{jt} + β_3 Treatment_i * Share_treated_{jt} + $X_i^{'} \varphi + \gamma_i + \delta_t + \varepsilon_{ijt}$

where, as before, *i* indexes individuals, *j* employment office, and *t* randomization month. *IES_attendance*_{ijt} is an indicator for reporting to the employment office twelve months after randomization; *Treatment*_i is an indicator that denotes whether jobseeker i was assigned to treatment; *Share_treated*_{jt} is the share of jobseekers assigned to treatment from the incoming flow in employment office *j* in month *t*; *X*_i is a vector of individual characteristics; γ_j are employment office fixed effects; and δ_t are month fixed effects. The coefficients of interest are β_2 and β_3 , which provide evidence on whether the share treated at the same office and in the same month is associated with the likelihood of reporting to the employment office twelve months after randomization for individuals in the control (β_2) or the treatment group ($\beta_2 + \beta_3$).

The results are presented in Appendix Table A16. Column (1) reports the effect of treatment on the probability of reporting to the employment office before the share of treated individuals is added into the model (a simple model that does not include β_2 or β_3). The estimate based on this extended sample and alternative model is similar in magnitude to that reported in Table 3, showing that the program reduced the probability of reporting to a labor office twelve months after randomization by 12.5 percentage points (s.e.=0.011). This is an important result because it shows that this alternative specification and an extended sample yield a similar treatment effect. The treatment coefficient changes little after we control for the share treated in the same office and month as reported in column (2). In column (3) we also introduce the interaction term between shared treated and the treatment indicator. Both coefficients are small and not significant, ruling out the possibility of externalities among the treated and the control group (or at least suggesting that if these externalities exist, they may have positive and negative effects that cancel each other out).

Appendix 3. Survey Questions for Assessment of Non-Cognitive Skills

In addition to standard demographic, employment, and earnings questions, both surveys (Wave 1 and Wave 2) included additional modules meant to measure respondents' non cognitive skills. For logistical reasons that limited survey length, not all modules were included in both surveys. In addition, as detailed below, some domains included only a selected number of items.

Job search self-efficacy module (Waves 1 and 2)

I will now read a series of statements. For each statement, please note whether you agree and whether you think it describes you accurately, using the following scale:

1—Strongly agree, 2—Agree, 3—Moderately agree, 4—Disagree, 5-Strongly disagree

- 1. I am confident in my ability to search for a job.
- 2. I am confident in my ability to use the internet in order to find a job.
- 3. I am confident in my ability to write a résumé.
- 4. I am confident in my ability to pass a job interview.

Source: Israel Employment Service

Work self-efficacy module (Waves 1 and 2)

I will now read a series of statements. For each statement, please note whether you agree and whether you think it describes you accurately, using the following scale:

1—Strongly agree, 2—Agree, 3—Moderately agree, 4—Disagree, 5-Strongly disagree

Thinking of my current or future work, I feel I will be able to...

- 1. Achieve goals that will be assigned.
- 2. Respect schedules and working deadlines.
- 3. Learn new working methods.
- 4. Concentrate all my energy on work.
- 5. Collaborate with other colleagues.
- 6. Have good relationships with my superiors.
- 7. Be courteous to customers.
- 8. Get to work on time.

Source: selected items from Pepe, Silvia J., et al., "Work Self-Efficacy Scale and Search for Work Self-efficacy Scale: A Validation study in Spanish and Italian Cultural Contexts." *Revista de Psicología del Trabajo y de las Organizaciones* 26.3 (2010): 201–210.

General self-efficacy module (Waves 1 and 2)

I will now read a number of statements. For each statement, please respond on a 5-point scale as to what extent it describes you.

1-Describes me very well, 2—Describes me well, 3—Describes me somewhat, 4—Doesn't describe me well, 5—Doesn't describe me at all

- 1. I can always manage to solve difficult problems if I try hard enough.
- 2. If someone opposes me, I can find the means and ways to get what I want.
- 3. It is easy for me to stick to my aims and accomplish my goals.
- 4. I can usually handle whatever comes my way.

Source: selected items in Schwarzer, R., and Jerusalem, M. (1995). "Generalized Self-Efficacy Scale," In J. Weinman, S. Wright, and M. Johnston, *Measures in Health Psychology: A User's Portfolio. Causal and Control Beliefs* (pp. 35-37). Windsor, UK: NFER-NELSON.

Grit Module (Wave 2)

I will now read a number of statements. For each statement, please respond on a 5-point scale as to what extent it describes you.

1—Describes me very well, 2—Describes me well, 3—Describes me somewhat, 4—Doesn't describe me well, 5—Doesn't describe me at all

- 1. New ideas and projects sometimes distract me from previous ones.
- 2. Setbacks don't discourage me.
- 3. I have been obsessed with a certain idea or project for a short time but later lost interest.
- 4. I am a hard worker.
- 5. I often set a goal but later choose to pursue a different one.
- 6. I have difficulty maintaining my focus on projects that take more than a few months to complete.
- 7. I finish whatever I begin.
- 8. I am diligent.

Items 1, 3, 5, and 6 are reverse-scored.

Source: "The Short Grit Scale," in Duckworth, Angela Lee, and Patrick D. Quinn, "Development and Validation of the Short Grit Scale (GRIT–S)." *Journal of Personality Assessment 91.2 (2009)*: 166–174.

Self-esteem module (Wave 2)

I will ask you to relate to a number of statements dealing with your general feelings about yourself. Please respond using the following 4-point scale as to how strongly you agree or disagree with each statement.

1—Strongly agree, 2—Agree, 3—Disagree, 4-Strongly disagree

- 1. On the whole, I am satisfied with myself.
- 2. At times I think I am no good at all.
- 3. I feel that I have a number of good qualities.
- 4. I am able to do things as well as most other people.
- 5. I feel I do not have much to be proud of.
- 6. I certainly feel useless at times.
- 7. I feel that I am a person of worth, at least on an equal plane with others.
- 8. I wish I could have more respect for myself.
- 9. All in all, I am inclined to feel that I am a failure.
- 10. I take a positive attitude toward myself.

Items 2, 5, 6, 8, and 9 are reverse-scored.

Source: "The Rosenberg Self-Esteem Scale" in Rosenberg, Morris, "Rosenberg Self-Esteem Scale (RSE)." *Acceptance and Commitment Therapy*. Measures Package 61.52 (1965): 18.

transferred					
Treated	0.005 (0.008)	More than 12 years of schooling	0.012 (0.017)		
Female	-0.003 (0.008)	Received income support months [-12;0]	-0.074*** (0.011)		
Age	-0.002*** (0.000)	Received income support months [-24;-11]	0.010 (0.009)		
Married	0.001 (0.012)	Received income support months [-36;-23]	-0.013 (0.011)		
Children	0.001	Months worked	-0.003		
	(0.002)	months [-12;0]	(0.002)		
Single parent	-0.032***	Months worked	-0.001		
	(0.011)	months [-24;-11]	(0.002)		
Immigrant	0.002	Months worked	0.001		
	(0.011)	months [-36;-23]	(0.002)		
Self-reported health limitation	-0.032***	Total earnings	0.000		
	(0.007)	months [-12;0]	(0.000)		
Arab	-0.012	Total earnings	-0.000		
	(0.014)	months [-24;-11]	(0.000)		
Ultra Orthodox	-0.004	Total earnings	0.000		
	(0.015)	months [-36;-23]	(0.000)		
12 years of schooling	0.001 (0.008)	Ν	6,744		

Table A1. Probability to stop reporting to the employment office before the randomization lists are
transferred

Notes: The table reports estimates from a linear probability model. The outcome is an indicator for stop reporting to the employment office before the randomization lists are transferred. Control variables include treatment status, individual's characteristics, and randomization unit fixed effects. * p < 0.05, ** p < 0.01, *** p < 0.001.

	Works (1)	Does not work and does not get income support (2)	Gets income support and does not work (3)
Income from work 12 months after randomization	3678	0	0
Income from work 12 months before randomization	1654	1004	638
Difference	2023	-1004	-638
Income support 12 months after randomization	331	0	1667
Income support 12 months before randomization	286	212	740
Difference	44	-212	928
Total Income 12 months after randomization	4008	0	1667
Total Income 12 months before randomization	1940	1216	1378
Difference	2068	-1216	290
Number of observations	1370	1060	618

Notes: The table reports a decomposition of program particpants' income 12 months before and after assignment to treatment according to their employment status 12 months after random assignemnt. Monetary values in real 2016 NIS.

Table A3a. Heterogeneous	Effects of	the	Program
--------------------------	------------	-----	---------

				of the ridgian	Ultra		
	N 4		1	A use le s	Orthodox	A == - 25	A 25
	Men	Women	Jews	Arabs	Jews	Age <35	Age >=35
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Reporting to employment office	-0.138**	-0.158**	-0.102**	-0.229**	-0.111**	-0.102**	-0.203**
	(0.024)	(0.023)	(0.017)	(0.032)	(0.047)	(0.023)	(0.027)
	0.347	0.410	0.305	0.466	0.378	0.290	0.493
Employed	0.062**	0.083**	0.039**	0.140**	0.065	0.054**	0.109**
	(0.023)	(0.019)	(0.019)	(0.019)	(0.044)	(0.019)	(0.020)
	0.391	0.289	0.432	0.227	0.351	0.380	0.276
Number of months employed	0.729**	0.899**	0.474**	1.462**	0.981**	0.716**	1.026**
	(0.215)	(0.131)	(0.141)	(0.159)	(0.386)	(0.155)	(0.165)
	3.745	2.950	4.461	2.048	3.401	3.775	2.705
Income from work	119.303	137.172**	47.625	318.729**	156.105	144.262	166.096*
(Including zeroes)	(130.904)	(62.308)	(85.633)	(86.227)	(171.893)	(101.781)	(92.010)
	1935.163	932.438	1799.845	873.418	1123.021	1501.937	1165.413
Cumulative income from work	1,730.610	1,854.732**	965.299	3,400.894**	2,932.931*	2,001.274**	2,003.713**
(Including zeroes)	(1136.883)	(490.095)	(705.443)	(772.319)	(1671.310)	(902.638)	(694.702)
	17417.234	8718.058	17060.232	7357.457	9854.810	13923.503	10434.754
Received Income support	-0.086**	-0.115**	-0.071**	-0.160**	-0.039	-0.083**	-0.123**
	(0.023)	(0.020)	(0.017)	(0.028)	(0.043)	(0.022)	(0.025)
	0.342	0.455	0.339	0.480	0.458	0.336	0.492
Income support payments	-149.996**	-183.391**	-136.940**	-232.385**	-117.990*	-140.148**	-199.517**
(Including zeroes)	(33.400)	(39.876)	(31.130)	(44.355)	(64.815)	(41.197)	(40.730)
	478.856	728.082	556.633	696.899	631.099	515.894	751.423
Cumulative income support	-1710.018**	-1914.658**	-1704.820**	-2250.202**	-1490.953**	-1559.621**	-2164.527**
(Including zeroes)	(285.972)	(415.274)	(312.237)	(451.096)	(740.744)	(377.436)	(389.080)
	7152.017	9975.647	8048.302	9606.670	8614.368	7609.141	10196.957
Total Income	-30.693	-46.219	-89.315	86.344	38.115	4.114	-33.421
(Including zeroes)	(131.042)	(69.374)	(84.041)	(103.340)	(172.272)	(118.080)	(87.931)
	2414.020	1660.521	2356.479	1570.317	1754.119	2017.831	1916.837
Total cumulative income	20.592	-59.926	-739.520	1,150.692	1,441.978	441.653	-160.814
(Including zeroes)	(1117.231)	(611.742)	(732.331)	(954.673)	(1748.272)	(1059.662)	(720.385)
	24569.250	18693.705	25108.535	16964.127	18469.178	21532.643	20631.711
Received other welfare	-0.007	-0.006	-0.009	-0.007	-0.008	-0.004	-0.019
payments (disability or UI or	(0.012)	(0.012)	(0.014)	(0.012)	(0.027)	(0.011)	(0.014)
other)	0.092	0.124	0.136	0.086	0.104	0.079	0.149
Number of observations	2,675	3,476	3,593	2,558	905	3,144	3,007

Notes: The table reports the program effect on different sub-populations. Controls include the relevant set from the main control list: sex, marital status, age, number of children, schooling level, indicators for new immigrant, single mothers, Arab, ultra-orthodox Jew, self-reported health limitations, vectors for employment, income from work and welfare history, and randomization unit fixed effects. Monetary values in real 2016 NIS. Control group means in italics. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

	No Self					
	Reported Health	Self Reported Health		Less Than 12 Years Of	12 years of	more than 12 years of
	Limitations	Limitations	Single Parents	Schooling	, schooling	, schooling
	(1)	(2)	(3)	(4)	(5)	(6)
Reporting to employment office	-0.110**	-0.238**	-0.169**	-0.205**	-0.119**	0.044
	(0.020)	(0.030)	(0.031)	(0.025)	(0.024)	(0.088)
	0.340	0.479	0.425	0.467	0.328	0.173
Employed	0.049**	0.144**	0.073**	0.105**	0.063**	0.053
	(0.016)	(0.022)	(0.035)	(0.021)	(0.020)	(0.078)
	0.374	0.238	0.358	0.260	0.376	0.547
Number of months employed	0.592**	1.502**	0.929**	1.170**	0.716**	-0.090
	(0.135)	(0.206)	(0.282)	(0.158)	(0.160)	(0.729)
	3.763	2.216	3.534	2.508	3.711	6.113
Income from work	31.544	433.296**	203.857	192.277**	166.580**	-284.022
(Including zeroes)	(86.427)	(103.874)	(135.130)	(82.856)	(83.121)	(867.399)
	1550.668	896.815	1213.928	1023.876	1413.231	3617.648
Cumulative income from work	961.157	4,392.611**	2,693.079**	2,703.320**	1,780.159**	-3666.443
(Including zeroes)	(755.428)	(774.413)	(1083.990)	(635.666)	(689.910)	(8500.292)
	14440.458	7624.496	11180.235	9079.598	12985.942	35009.859
Received Income support	-0.085**	-0.153**	-0.084**	-0.144**	-0.084**	0.091
	(0.019)	(0.023)	(0.035)	(0.022)	(0.020)	(0.088)
	0.387	0.456	0.455	0.492	0.355	0.167
Income support payments	-140.091**	-242.835**	-158.917*	-213.227**	-158.266**	176.674
(Including zeroes)	(35.238)	(39.000)	(86.292)	(39.188)	(35.609)	(163.567)
	604.770	670.603	894.952	754.050	540.825	267.320
Cumulative income support	-1574.414**	-2456.417**	-2270.409**	-2447.090**	-1634.693**	1,525.121
(Including zeroes)	(343.518)	(423.796)	(956.902)	(360.131)	(344.355)	(1579.506)
	8547.170	9393.063	12170.298	10221.449	7938.194	4389.525
Total Income	-108.548	190.460*	44.940	-20.950	8.314	-107.348
(Including zeroes)	(94.300)	(110.324)	(141.113)	(87.714)	(85.536)	(860.100)
	2155.438	1567.418	2108.880	1777.926	1954.056	3884.968
Total cumulative income	-613.258	1,936.194**	422.670	256.230	145.466	-2141.323
(Including zeroes)	(839.169)	(862.647)	(1378.514)	(715.668)	(748.589)	(8251.045)
	22987.629	17017.559	23350.533	19301.047	20924.137	39399.383
Received other welfare payments	-0.011	-0.008	-0.031	-0.015	-0.005	-0.094**
(disability or UI or other)	(0.009)	(0.019)	(0.031)	(0.013)	(0.012)	(0.048)
	0.079	0.183	0.283	0.114	0.111	0.048)
Number of observations	4,066	2,085	1,258	2,625	3,215	311

Notes: The table reports the program effect on different sub-populations. Controls include the relevant set from the main control list: sex, marital status, age, number of children, schooling level, indicators for new immigrant, single mothers, Arab, ultra-orthodox Jew, self-reported health limitations, vectors for employment, income from work and welfare history, and randomization unit fixed effects. Monetary values in real 2016 NIS. Control group means in italics. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

Table A3c. Heterogeneous Effects of the Program

	No Recent Income Support	Recent Income Support History	No Recent Employment History	Recent Employment History	Local Unemploment rate < 7.5%	Local Unemploment rate >=7.5%
	History (1)	(2)	(3)	(4)	(5)	(6)
Reporting to employment office	-0.127** (0.025) 0.314	-0.170** (0.021) 0.464	-0.206** (0.030) 0.491	-0.115** (0.018) 0.304	-0.117** (0.020) 0.300	-0.173** (0.029) 0.438
Employed	0.046* (0.024) 0.373	0.112** (0.018) 0.284	0.093** (0.018) 0.166	0.068** (0.020) 0.455	0.056** (0.024) 0.415	0.095** (0.018) 0.282
Number of months employed	0.509** (0.185) 3.782	1.191** (0.163) 2.704	1.102** (0.155) 1.378	0.727** (0.176) 4.701	0.481** (0.162) 4.368	1.122** (0.155) 2.638
Income from work (Including zeroes)	51.285 (110.641) 1603.272	259.203** (85.647) 1052.078	121.553* (69.276) 612.126	193.397** (95.343) 1895.182	70.504 (103.866) 1763.493	206.718** (79.532) 1082.507
Cumulative income from work (Including zeroes)	1,305.442 (966.799) 14894.933	2,599.766** (705.347) 9349.802	2,121.021** (628.718) 4590.745	1,949.087** (799.037) 18081.299	1,053.616 (792.328) 16887.566	2,472.658** (725.886) 9515.750
Received Income support	-0.090** (0.023) 0.270	-0.118** (0.020) 0.566	-0.129** (0.022) 0.537	-0.091** (0.021) 0.312	-0.062** (0.019) 0.324	-0.135** (0.026) 0.467
Income support payments (Including zeroes)	-138.395** (38.943) 392.579	-202.724** (36.636) 890.352	-217.213** (39.369) 837.237	-143.504** (35.774) 466.647	-134.578** (38.480) 520.581	-193.341** (43.362) 699.210
Cumulative income support (Including zeroes)	-1504.281** (356.120) 5727.101	-2224.101** (344.917) 12323.116	-2073.753** (449.648) 11118.897	-1761.189** (353.713) 7083.726	-1729.152** (388.979) 7696.883	-1942.050** (400.936) 9598.610
Total Income (Including zeroes)	-87.110 (117.751) 1995.850	56.478 (84.294) 1942.429	-95.659 (75.653) 1449.363	49.893 (100.255) 2361.828	-64.075 (97.507) 2284.073	13.377 (98.527) 1781.717
Total cumulative income (Including zeroes)	-198.838 (1096.140) 20622.033	375.666 (774.294) 21672.918	47.267 (739.422) 15709.643	187.898 (910.179) 25165.023	-675.536 (806.437) 24584.449	530.608 (932.220) 19114.359
Received other welfare payments (disability or UI or other)	-0.023* (0.013) 0.131	0.002 (0.011) 0.089	-0.008 (0.015) 0.107	-0.011 (0.012) 0.114	-0.021 (0.017) 0.146	-0.001 (0.009) 0.092
Number of observations	3,002	3,149	2,565	3,586	2,632	3,415

Notes: The table reports the program effect on different sub-populations. Recent income support history refers to individuals who had at least one spell of income support during the two years prior to randomization. Recent employment history refers to individuals who had at least one employment spell during the two years prior to randomization. Controls include the relevant set from the main control list: sex, marital status, age, number of children, schooling level, indicators for new immigrant, single mothers, Arab, ultra-orthodox Jew, self-reported health limitations, vectors for employment, income from work and welfare history, and randomization unit fixed effects. Monetary values in real 2016 NIS. Control group means in italics. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

	(1)	(2)	n into the Survey (1)	(2)
Treated	0.014 (0.015)	-0.084 (0.073)	Treated * Female	0.023 (0.035)
Female	0.014 (0.018)	0.005 (0.025)	Treated * Age	0.001 (0.002)
Age	0.001 (0.001)	0.001 (0.001)	Treated * Married	-0.014 (0.040)
Married	0.038* (0.020)	0.041 (0.027)	Treated * Children	0.009 (0.008)
Children	0.003 (0.005)	-0.001 (0.006)	Treated * Single parent	0.017 (0.043)
Single parent	0.026 (0.019)	0.013 (0.029)	Treated * Immigrant	-0.053 (0.036)
mmigrant	-0.063*** (0.021)	-0.032 (0.027)	Treated * Self-reported health limitation	-0.075** (0.030)
Self-reported health limitation	0.049*** (0.015)	0.087*** (0.023)	Treated * Arab	0.095*** (0.035)
Arab	0.016 (0.021)	-0.026 (0.024)	Treated * Ultra Orthodox	0.017 (0.051)
Jltra Orthodox	0.084*** (0.026)	0.076* (0.044)	Treated * 12 years of schooling	0.024 (0.029)
2 years of schooling	0.095*** (0.016)	0.082*** (0.022)	Treated * More than 12 years of schooling	0.086 (0.065)
Nore than 12 years of schooling	0.194*** (0.031)	0.145*** (0.048)	Treated * Received income support months [-12;0]	-0.011 (0.037)
Received income support nonths [-12;0]	-0.001 (0.019)	0.008 (0.025)	Treated * Received income support months [-24;-11]	0.080 (0.050)
Received income support nonths [-24;-11]	0.042** (0.020)	0.000 (0.034)	Treated * Received income support months [-36;-23]	-0.033 (0.044)
Received income support nonths [-36;-23]	-0.016 (0.020)	0.004 (0.033)	Treated * Months worked months [-12;0]	0.002 (0.007)
Nonths worked nonths [-12;0]	-0.001 (0.004)	-0.003 (0.006)	Treated * Months worked months [-24;-11]	-0.003 (0.008)
Nonths worked nonths [-24;-11]	0.002 (0.003)	0.004 (0.005)	Treated * Months worked months [-36;-23]	-0.005 (0.005)
Nonths worked nonths [-36;-23]	0.006* (0.003)	0.009** (0.004)	Treated * Total earnings months [-12;0]	-0.001 (0.016)
⁻ otal earnings nonths [-12;-0]	-0.001 (0.008)	0.000 (0.011)	Treated * Total earnings months [-24;-11]	0.001 (0.016)
Fotal earnings nonths [-24;-11]	0.005 (0.007)	0.004 (0.009)	Treated * Total earnings months [-36;-23]	0.002 (0.010)
⁻ otal earnings nonths [-36;-23]	-0.001 (0.005)	-0.001 (0.007)	Treated * First survey pop. sample	0.045 (0.028)
First survey pop. sample	0.350*** (0.021)	0.333*** (0.016)	Treated * Claimant type	-0.006 (0.034)
			F-Stat for joint significance	4.875
Notes: The table reports the probabil	lity of survey resp	onse as a	P-value	<0.001
function of personal characteristics a	nd program assign	nment.	N 6,713	6,713

conditional on randomization unit fixed effects. The F-stat is for a test of joint significance of treatment and all interactions with treatment.

Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

Table A5. Estimation of Survey Weights - Probability of Inclusion into Survey Sample	Table A5.	Estimation	of Survey	Weights -	Probability	of Inclusion	into Surve	/ Sample
--	-----------	------------	-----------	-----------	-------------	--------------	------------	----------

Treated	0.066 (0.295)	Treated * Female	0.086 (0.123)
Female	0.029 (0.090)	Treated * Age	-0.001 (0.007)
Age	0.004 (0.005)	Treated * Married	-0.070 (0.170)
Married	0.179 (0.123)	Treated * Children	0.034 (0.036)
Children	-0.001 (0.025)	Treated * Single parent	0.104 (0.182)
Single parent	0.020 (0.135)	Treated * Immigrant	-0.177 (0.156)
Immigrant	-0.153 (0.116)	Treated * Self-reported health limitation	-0.276** (0.123)
Self-reported health limitation	0.313*** (0.090)	Treated * Arab	0.413*** (0.151)
Arab	-0.084 (0.126)	Treated * Ultra Orthodox	0.089 (0.199)
Ultra Orthodox	0.328** (0.158)	Treated * 12 years of schooling	0.134 (0.121)
12 years of schooling	0.279*** (0.088)	Treated * More than 12 years of schooling	0.257 (0.266)
More than 12 years of schooling	0.646*** (0.194)	Treated * Received income support months [-12;0]	-0.179 (0.132)
Received income support months [-12;0]	0.199* (0.103)	Treated * Received income support months [-24;-11]	0.402** (0.188)
Received income support months [-24;-11]	-0.028 (0.141)	Treated * Received income support months [-36;-23]	-0.188 (0.186)
Received income support months [-36;-23]	-0.013 (0.139)	Treated * Months worked months [-12;0]	0.016 (0.029)
Months worked months [-12;0]	-0.016 (0.021)	Treated * Months worked months [-24;-11]	-0.030 (0.030)
Months worked months [-24;-11]	0.027 (0.022)	Treated * Months worked months [-36;-23]	-0.014 (0.025)
Months worked months [-36;-23]	0.030* (0.018)	Treated * Total earnings months [-12;0]	-0.019 (0.064)
Total earnings months [-12;-0]	0.013 (0.046)	Treated * Total earnings months [-24;-11]	0.032 (0.057)
Total earnings months [-24;-11]	0.004 (0.040)	Treated * Total earnings months [-36;-23]	0.001 (0.044)
Total earnings months [-36;-23]	-0.010 (0.031)	Constant	-0.951** (0.406)
		N	6,117

Notes: The table reports the estimates of a logistic regression that estimates likelihood of survey response as a function of personal characteristics and program assignment, conditional on randomization unit fixed effects. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

	treated (1)	T-C (2)		treated (1)	T-C (2)
Female	0.54	-0.024 (0.023)	Months worked months [-12;0]	2.84	-0.061 (0.199)
Age	34.56	0.129 (0.492)	Months worked months [-24;-11]	3.96	0.098 (0.242)
Married	0.47	0.007 (0.020)	Months worked months [-36;-23]	4.31	0.223 (0.254)
Children	2.00	0.014 (0.092)	Total earnings months [-12;0]	9846	150 (696)
Single parent	0.22	0.002 (0.021)	Total earnings months [-24;-11]	16341	1220 (1294)
Immigrant	0.20	-0.018 (0.019)	Total earnings months [-36;-23]	18284	1100 (1536)
Self-reported health limitation	0.36	0.009 (0.021)	Total income support months [-12;0]	6106	140 (424)
Arab	0.35	-0.002 (0.014)	Total income support months [-24;-11]	4040	250 (389)
Ultra Orthodox	0.19	0.025* (0.013)	Total income support months [-36;-23]	3263	90 (318)
Less than 12 years of schooling	0.39	-0.033 (0.024)	Months since random assignm	13.60	-0.464 (0.000)
12 years of schooling	0.56	0.032 (0.024)	F-Stat for joint significance	0.693	
More than 12 years of schooling	g 0.05	0 (0.010)	P-value Number of observations	0.835 1,702	3,044
Received income support months [-12;0]	0.52	-0.015 (0.028)			
Received income support months [-24;-11]	0.28	0.003 (0.021)			
Received income support months [-36;-23]	0.24	0.004 (0.019)			

Table A6. Descriptive Statistics and Balancing Tests - Survey Sample

Notes: The table reports the average characteristics of treatment group (column 1) alongside the estimated difference with the control group, conditional on randomization unit fixed effects (column 2). The sample is restricted on survey respondent. The reported F statistic tests the joint significance of all covariants in a linear probability model that predicts treatment status conditional on randomization unit fixed effects. Observations are weighted by survey weights. Monetary values in real 2016 NIS. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

Table A7. Main Results Based on Survey Sample						
	Impact after 12 months					
	(1)					
Reporting to employment office	-0.157***					
	(0.023)					
	0.409					
Employment	0.089***					
	(0.023)					
	0.355					
Income from work	119					
(Including zeroes)	(107)					
	1,477					
Cumulative income from work	1510					
(Including zeroes)	(973)					
	13,501					
Received Income support	-0.083***					
	(0.023)					
	0.423					
Income support payments	-131***					
(Including zeroes)	(43)					
	621					
Cumulative income support	-1364***					
(Including zeroes)	(469)					
	8,776					
Total Income	-12					
(Including zeroes)	(111)					
	2,098					
Total cumulative income	146					
(Including zeroes)	(1037)					
	22,276					
Received other welfare payments	-0.003					
(disability or UI or other)	(0.013)					
	0.109					
N	3,064					

Notes: The table reports the program effect on participants' outcomes. The sample is restricted to survey respondents. All regressions control for the same set of covariates reported in Table 3 and include randomization unit fixed effects. Observations are weighted by survey weights. Monetary values in real 2016 NIS. Control group means in italics. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

Table A8. Reliability Coefficients of Survey	/ Constructs					
Item	Obs (1)	Sign (2)	Item-test correlation (3)	Item-rest correlation (4)	Average interitem covariance (5)	Alpha (6)
Search efficacy					0.612	0.863
I am confident in my abilities to search for a job	2750	+	0.835	0.689	0.623	0.832
I am confident in my ability to use the internet in order to find a job	2725	+	0.816	0.660	0.643	0.844
I am confident in my ability to write a resume	2775	+	0.864	0.738	0.591	0.813
I am confident in my ability to pass a job interview	2701	+	0.861	0.737	0.591	0.813
Work self-efficacy					0.760	0.962
Achieve goals that will be assigned	2729	+	0.875	0.832	0.766	0.958
Respect schedules and working deadlines	2756	+	0.889	0.850	0.761	0.957
Learn new working methods	2719	+	0.862	0.816	0.769	0.959
Concentrate all energy on work	2738	+	0.887	0.848	0.761	0.957
Collaborate with other colleagues	2747	+	0.912	0.882	0.752	0.955
Have good relationships with my superiors	2733	+	0.912	0.881	0.753	0.955
Be courteous to customers	2711	+	0.901	0.867	0.756	0.956
Get to work on time	2748	+	0.886	0.847	0.762	0.957
General self-efficacy					0.609	0.862
I can always manage to solve difficult problems if I try hard enough	2794	+	0.850	0.713	0.604	0.821
If someone opposes me, I can find the means and ways to get what I want	2753	+	0.850	0.717	0.600	0.818
It is easy for me to stick to my aims and accomplish my goals	2785	+	0.831	0.682	0.624	0.833
I can usually handle whatever comes my way	2757	+	0.842	0.704	0.608	0.823

					Average	
Item			Item-test	Item-rest	interitem	
	Obs	Sign	correlation	correlation	covariance	Alpha
	(1)	(2)	(3)	(4)	(5)	(6)
Grit					0.137	0.559
New ideas and projects sometimes distract me from previous ones (reversed)	831	+	0.429	0.172	0.151	0.555
Setbacks don't discourage me	924	+	0.368	0.100	0.166	0.583
I have been obsessed with a certain idea or project for a short time but later lost interest (reversed)	848	+	0.533	0.299	0.130	0.511
I am a hard worker	889	+	0.453	0.197	0.148	0.549
I often set a goal but later choose to pursue a different one (reversed)	866	+	0.476	0.227	0.140	0.533
I have difficulty maintaining my focus on projects that take more than a few months to complete (reversed)	838	+	0.572	0.356	0.122	0.494
I finish whatever I begin	938	+	0.609	0.388	0.117	0.481
I am diligent	929	+	0.609	0.384	0.120	0.488
Self esteem					0.268	0.785
On the whole, I am satisfied with myself	976	+	0.642	0.492	0.263	0.763
At times I think I am no good at all (reversed)	947	+	0.581	0.432	0.268	0.768
I feel that I have a number of good qualities	955	+	0.637	0.501	0.261	0.761
I am able to do things as well as most other people	950	+	0.647	0.513	0.259	0.758
I feel I do not have much to be proud of (reversed)	872	+	0.410	0.246	0.294	0.790
I certainly feel useless at times (reversed)	877	+	0.612	0.475	0.262	0.762
I feel that I am a person of worth, at least on an equal plane with others	919	+	0.572	0.429	0.270	0.769
I wish I could have more respect for myself (reversed)	879	+	0.476	0.317	0.285	0.782
All in all, I am inclined to feel that I am a failure (reversed)	853	+	0.653	0.532	0.257	0.757
I take a positive attitude toward myself	933	+	0.637	0.503	0.259	0.759

Table A8. (cont.) Reliability Coefficients of Survey Constructs

Notes: The table reports the inter-item correlations and Cronbach's alpha for the different non-cognitive domains included in the survey.

Table A9. Correlations Between Survey Constructs										
	Job search self efficacy score (1)	Work self efficacy score (2)	Self efficacy score (3)	Grit score (4)	Self esteem score (5)					
Job search self efficacy score	1.000	0.636	0.518	0.364	0.436					
Work self efficacy score	0.636	1.000	0.603	0.447	0.477					
Self efficacy score	0.518	0.603	1.000	0.464	0.542					
Grit score	0.364	0.447	0.464	1.000	0.517					
Self esteem score	0.436	0.477	0.542	0.517	1.000					

Notes: The table reports the variance-covariance matrix of the standardized aggregate non-cognitive scores in the survey sample.

	N (1)	Treatment effect (2)
I am confident in my abilities to search for a job	2750	0.042 (0.048)
I am confident in my ability to use the internet in order to find a job	2725	0.069* (0.038)
I am confident in my ability to write a resume	2775	0.054 (0.041)
I am confident in my ability to pass a job interview	2701	0.068 (0.044)

Notes: The table reports the program effect on participants' standardized job search self-efficacy items. All regressions control for the same set of covariates reported in Table 3 and include also survey month and randomization unit fixed effects. Observations are weighted by survey weights. Standard errors clustered at the randomization unit level in parentheses. * p < 0.05, ** p < 0.01, *** p < 0.001.

Table A10. Program Effect on Search Efficacy

	Ν	Treatment effect
I Feel I can	(1)	(2)
Achieve goals that will be assigned	2729	0.060 (0.044)
Respect schedules and working deadlines	2756	0.072* (0.042)
Learn new working methods	2719	0.072* (0.044)
Concentrate all energy on work	2738	0.100** (0.047)
Collaborate with other colleagues	2747	0.107** (0.045)
Have good relationships with my superiors	2733	0.073 (0.051)
Be courteous to customers	2711	0.103** (0.048)
Get to work on time	2748	0.094** (0.047)

Table A11. Program Effect on Work Self-Efficacy

Notes: The table reports the program effect on participants' standardized work self-efficacy items. All regressions control for the same set of covariates reported in Table 3 and include also survey month and randomization unit fixed effects. Observations are weighted by survey weights. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

	N	Treatment effect
	(1)	(2)
I can always manage to solve difficult problems if I try hard enough	2794	-0.064 (0.051)
If someone opposes me, I can find the means and ways to get what I want	2753	0.084 (0.052)
It is easy for me to stick to my aims and accomplish my goals	2785	-0.029 (0.055)
I can usually handle whatever comes my way	2757	0.030 (0.044)

Table A12. Program Effect on Self-Efficacy

Notes: The table reports the program effect on participants' standardized general self-efficacy items. All regressions control for the same set of covariates reported in Table 3 and include also survey month and randomization unit fixed effects. Observations are weighted by survey weights. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

	N (1)	Treatment effect (2)
New ideas and projects sometimes distract me from previous ones (reversed)	831	-0.003 (0.089)
Setbacks don't discourage me	924	0.098 (0.078)
I have been obsessed with a certain idea or project for a short time but later lost interest (reversed)	848	-0.128 (0.080)
I am a hard worker	889	0.097 (0.083)
I often set a goal but later choose to pursue a different one (reversed)	866	-0.153 (0.093)
I have difficulty maintaining my focus on projects that take more than a few months to complete (reversed)	838	-0.014
		(0.098)
I finish whatever I begin	938	-0.141 (0.085)
I am diligent	929	0.056 (0.069)

Table A13. Program Effect on Grit

Notes: The table reports the program effect on participants' standardized grit items. All regressions control for the same set of covariates reported in Table 3 and include also survey month and randomization unit fixed effects. Observations are weighted by survey weights. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

	Ν	Treatment effect
	(1)	(2)
On the whole, I am satisfied with myself	976	-0.018
		(0.080)
At times I think I am no good at all (reversed)	947	0.046
		(0.078)
I feel that I have a number of good qualities	955	0.093
		(0.099)
I am able to do things as well as most other people	950	0.082
		(0.091)
I feel I do not have much to be proud of (reversed)	872	-0.000
		(0.095)
I certainly feel useless at times (reversed)	877	-0.011
		(0.076)
I feel that I am a person of worth, at least on an equal plane with	919	0.124
others		(0.109)
I wish I could have more respect for myself (reversed)	879	0.167**
		(0.079)
All in all, I am inclined to feel that I am a failure (reversed)	853	0.016
		(0.080)
I take a positive attitude toward myself	933	0.091
		(0.088)

Notes: The table reports the program effect on participants' standardized self-esteem items. All regressions control for the same set of covariates reported in Table 3 and include also survey month and randomization unit fixed effects. Observations are weighted by survey weights. Standard errors clustered at the randomization unit level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

Table A14. Program Effect on Self-Esteen		Table A14.	Program	Effect on	Self-Esteem
--	--	------------	---------	-----------	-------------

			Tad	e AIS. Balanci	ng rests by	Share freated	a in Employm	ent Office				
							Self-					
							reported			Less than		More than
				Number of	Single		health		Ultra	12 years of	12 years of	12 years of
	Female (1)	Age (2)	Married (3)	children (4)	parent (5)	Immigrant (6)	limitation (7)	Arab (8)	Orthodox (9)	schooling (10)	schooling (11)	schooling (12)
Share treated	0.006 (0.042)	-0.026 (0.823)	-0.028 (0.043)	-0.111 (0.201)	0.030 (0.032)	0.044 (0.039)	-0.066* (0.037)	-0.031 (0.031)	-0.032 (0.026)	-0.059 (0.050)	0.062 (0.050)	-0.002 (0.021)
Treated	-0.003 (0.022)	0.264 (0.360)	-0.008 (0.020)	0.128 (0.093)	0.030** (0.015)	-0.006 (0.020)	-0.001 (0.020)	0.009 (0.014)	-0.006 (0.013)	0.002 (0.021)	0.011 (0.022)	-0.013 (0.013)
Treated * Share treated	-0.026 (0.053)	-0.625 (0.944)	-0.014 (0.055)	-0.184 (0.236)	-0.041 (0.039)	-0.018 (0.050)	0.060 (0.049)	0.006 (0.041)	0.021 (0.036)	0.004 (0.058)	-0.026 (0.059)	0.021 (0.029)
Ν	16,635	16,635	16,635	16,635	16,635	16,635	16,635	16,635	16,635	16,635	16,635	16,635

Table A15. Balancing Tests by Share Treated in Employment Office

Notes: The table reports the association between the share of monthly treated individuals in each employment office and individuals' characteristics. Controls include employment office and month fixed effects. Standard errors clustered at the employment-office-month level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.

	Attendance at the employment office 12 months after random assignment						
	(1)	(2)	(3)				
Treatment	-0.125***	-0.121***	-0.141***				
	(0.011)	(0.011)	(0.022)				
Share Treated		-0.052	-0.078				
		(0.039)	(0.051)				
Share Treated X Treatment			0.057				
			(0.059)				
Ν	13,058	13,058	13,058				

Table A16. The Relationship Between Share Treated and Attendance at the Employment Office

Notes: The table reports the probability to report to the employment office 12 months after random assignment as a function of treatment status, the share of monthly treated individuals at the employment office and the interaction between both variables. All regressions control for the same set of covariates reported in Table 3 and include also employment office and month fixed effects. Standard errors clustered at the employment-office-month level in parentheses.* p < 0.05, ** p < 0.01, *** p < 0.001.